SUSTAINABLE

LAB

# A GREEN INTEREST RATE FOR THE EUROZONE

Evaluating the design choices

#### In this paper

The current high interest rates slow down the energy transition, making the Eurozone more vulnerable to fossilflation. With a green interest rate programme, the ECB can offset this effect. In this paper we analyse the most relevant design choices for such a green interest rate programme, like how to define 'green' and the structure and size of such a programme. Using data from the 2023 Taxonomy-alignment reports of banks we find that a green rate can accelerate the energy transition throughout the Eurozone against modest costs.

**Stanislas Jourdan, Rens van Tilburg, Aleksandar Simić, Brenda Kramer and Gaston Bronstering** October 2024



#### Colophon

Utrecht, October 2024.

The Sustainable Finance Lab (SFL) is an academic think tank whose members are mostly professors from different universities in the Netherlands. The aim of the SFL is a stable and robust financial sector that contributes to an economy that serves humanity without depleting its environment. To this end the SFL develops ideas and provides a platform to discuss them, thus bridging science and practice.

The authors are grateful to all members of the Sustainable Finance Lab for their input, in particular Dirk Schoenmaker, Clemens Kool, Mark Sanders, Dirk Bezemer and Friedemann Polzin. We also thank Pierre Monnin, Alexander Barkawi, Alain Grandjean and the participants of various workshops with national central banks and researchers for their comments. We further thank Pauline Simon for her research assistance and support with data collection, and Thomas Veenstra for his support with editing and readability. Finally, we thank Sunrise Project and the Laudes Foundation for their financial support.

#### **Policy Paper**

Sustainable Finance Lab publishes different types of publications. This is a policy paper. Policy papers are reports produced by SFL members or employees that contain specific proposals and recommendations for the financial sector or policy makers. The views expressed in this publication are those of the authors and do not necessarily reflect those of all members of the Sustainable Finance Lab.

2

# CONTENT

Summary		4
Glossary		7
1.	Introduction	8
2.	How the ECB's monetary policy works against price stability and EU energy transition	10
3.	Background and policy discussion	17
4.	Definition of 'green': Use of the EU Taxonomy	26
5.	Structure and size of a Green TLTRO	33
6.	Other implementation aspects	41
7.	Design choices and policy trade-offs	45
8.	Preliminary programme simulations	52
9.	Conclusion	59
An	Annex I: Figures from the dataset	
Bibliography		64

4

# SUMMARY

The EU's quest for strategic autonomy and energy decarbonation requires a more rapid transition away from fossil fuels and massive investments in renewable energy sources, energy efficiency, grid interconnectivity and storage. On the upside, the energy transition would also benefit price stability prospects, as a clean energy mix will likely result in lower and more stable energy prices. Under its primary and secondary objective, the ECB has the mandate to support these EU priorities.

The current high interest rate environment has slowed down the transition. As a result of their cost structure, green investments are more sensitive to interest rate changes than fossil fuel ones. They are, therefore, disproportionately hit by the ECB's high interest rate policy. This policy contradiction slows down the green transition in the EU, which ultimately poses risks to price stability.

To address this contradiction, the ECB could introduce a green interest rate, in the form of a green targeted longer term refinancing operation (Green TLTROs). A Green TLTROs would provide banks with a lower green interest rate under conditions that incentivizes them to extend more green loans at cheaper rate.

The obstacles against the introduction of a Green TLTROs identified by the ECB during its 2020-2021 Strategy review are currently being overcome. As banks and corporations are starting to report on their alignment with the EU Taxonomy, reliable data on green lending is now available for central banks. Moreover, the ECB has announced that it will consider climate change when introducing new structural refinancing operations in the next few years.

Coincidentally, as the ECB's monetary policy stance is expected to become more accommodative in the coming months, the ECB should seize this momentum to introduce a separate interest rate for green activities. Such initiative would also come at a timely moment to support the delivery of the EU's climate targets for 2030.

Designing a Green TLTRO is a delicate exercise. Such a policy should be created to stimulate green lending, while not jeopardizing the conduct of monetary policy. It should also be operationally feasible and preserve the level playing field between banks. This paper aims to clarify the design choices to be made in the Eurosystem context. The EU Taxonomy of sustainable activities provides a broad and robust definition of green lending, and the related mandatory disclosure for banks and corporations will greatly facilitates the implementation of Green TLTROs. In addition, the ECB could decide to adopt a more "selective" Taxonomy approach that could help to prioritize specific segments of the EU Taxonomy that are more relevant for the conduct of price stability.

The ECB could design the green rate in two complementary ways. First, participating banks could benefit from an initial "basic" interest rate reduction on the amount of newly originated Taxonomy-aligned lending during the previous year. An additional "bonus" rate reduction could be granted for banks reaching pre-determined lending targets, for instance based on the Green Asset Ratio. We advise against introducing a penalty rate on dirty lending at this stage. Based on the literature, we find that a green spread of 100 to 200bps would already have tangible effects in stimulating renewable investments.

We built a dataset based on the Taxonomy-alignment reports of 73 banks for 2023 representing 80% of Eurozone banking assets. This dataset enables us to estimate the amount of Taxonomy-aligned assets stock of €400-500 billion. If banks had to increase their Green Asset Ratios by 2 pp per year in order to qualify for the green bonus rate, the size of the Green TLTRO could reach an annual size of between €317-378 billion. This order of magnitude is in line with estimations of the size of the green investment gap in Europe, and falls within the possible envelope of the ECB's future structural operations.

We identify several options enabling the ECB to control the size of the Green TLTRO programme to fit its monetary policy stance. If needed, the ECB could reduce the borrowing allowances of banks, for instance by taking a selective approach in defining which EU Taxonomy activities would be eligible under the programme. In the extreme case the ECB can put a cap on the size of the whole programme.

Due to negative difference between the green interest rate and the ECB's deposit facility rate, a Green TLTRO would bear a cost on the Eurosystem's balance sheet. Under conservative assumptions, we estimate these losses at around €5 billion per year on the Eurosystem's balance sheet. In comparison, the Eurosystem is expected to make more than €160 billion of losses in 2022-2027.

# GLOSSARY

ΔΡΡ	Asset Purchase Programmes
CERE	Carbon Emission Reduction Facility
CSRD	Corporate Sustainability Reporting Directive
DFR	Deposit Facility Rate
DNR	De Nederlandsche Bank (Dutch central bank)
DNSH	Do No Significant Harm
FBA	European Banking Authority
FCB	European Central Bank
FONIA	Euro OverNight Index Average
EPBD	Energy Performance of Buildings Directive
FPC	Energy Performance Certificate
ESCB	European System of Central Banks
ESG	Environmental, Social and Governance
GAR	Green Asset Ratio
GDP	Gross Domestic Product
HICP	Harmonised Index of Consumer Prices
IEA	International Energy Agency
IMF	International Monetary Fund
IPCC	Intergovernmental Panel on Climate Change
KPI	Key Performance Indicator
LCOE	Levelized Cost of Electricity
LTRO	Long-Term Refinancing Operation
MRO	Main Refinancing Operation
NACE	Nomenclature statistique des activités économiques dans la
	Communauté européenne (Statistical classification of economic
	activities in the European Community)
NFRD	Non-Financial Reporting Directive
NGFS	Network for Greening the Financial System
NVDE	Nederlandse Vereniging Duurzame Energie (Netherlands
	Association for Renewable Energy)
NSFR	Net Stable Funding Ratio
PBOC	People's Bank of China
RED	Renewable Energy Directive
SAFE	Survey on the Access to Finance of Enterprises
SSM	Single Supervisory Mechanism
TFEU	Treaty on the Functioning of the European Union
TLTRO	Targeted Long-Term Refinancing Operation

# 1. INTRODUCTION

The unprecedented inflationary crisis of 2021-2024, resulting from the economic restart after the Covid19 pandemic and the Russian invasion in Ukraine, has unveiled the dangerous dependency of the EU economy on imported fossil fuels. "Fossilflation" emerged as a term for the inflation driven primarily by increased prices of imported fossil energy like oil and gas in the Eurozone (Schnabel, 2022). In response to the spike of inflation, the European Central Bank (ECB) intervened by raising interest rates in 2022.

As pointed out in the recent report by Mario Draghi, the best solution against fossilflation would be to accelerate the energy transition, by boosting investments into energy decarbonation such as renewable energy sources, energy efficiency, grid interconnectivity and storage (Draghi, 2024). A rapid energy transition would result in cheaper and more stable energy prices, which would also improve price stability.

However, the ECB's monetary tightening simultaneously slows down the energy transition, as it worsens the financing conditions for specific sectors such as renewable energy investments, energy efficiency of buildings and other industries that are more capital intensive than the fossil fuel sector. Thus, the ECB's monetary policy is arguably working against the EU's economic priorities, like strategic autonomy and the implementation of the EU's Green Deal.

A wide range of academics, business organisations and think tanks have suggested a green interest rate (also referred to as "dual rates" or "green targeted longer term refinancing operation" (Green TLTROs)) in order to support the financing of transition investments. Unlike Japan and China, the ECB has so far rejected this concept, in particular due to operational feasibility challenges, such as the lack of reliable data on green lending. Moreover, the ECB was reluctant to envisage the introduction dual rates under a restrictive policy environment as it was concerned it could blur its monetary policy stance.

However, the ECB started to cut interest rates in June 2024 and is expected to continue doing so in coming months, as inflation recedes in the Eurozone. Moreover, significant progress in the use of the EU Taxonomy by banks and corporates may facilitate the implementation of a green interest rate policy. Thus, the time may have come to revisit how a Green TLTRO programme could be introduced in practice. The goal of this paper is to evaluate the design options for a green interest rate in the Eurosystem policy framework. The paper does not intend to propose one way to introduce a green interest rate, but rather to guide policymakers and other stakeholders in their own thinking and develop more detailed proposals.

The report is structured as follows. Chapter 2 starts with an explanation of the two-way relationship between the ECB's monetary policy and the EU's energy transition. Chapter 3 reviews the recent policy discussion on the idea of a green interest rate. Chapter 4 discusses to what extent the implementation of the EU Taxonomy regulation can address the need for a clear and robust definition of green lending, and whether reliable data is sufficiently available. Chapter 5 describes how the level of green interest rate could be calculated. Chapter 6 discusses other policy parameters such as collateral requirements and the maturity of loans. In Chapter 7, we compare the advantages and disadvantages of the various policy parameters discussed and f highlight some of the policy trade-offs involved. For illustrative purposes, in chapter 8 we analyse the potential scenario of a green interest rate of up to 150 basis points lower than the ECB's key rate from 2025 until 2030. Based on a unique dataset of EU Taxonomy-alignment from a sample of European banks, we provide a preliminary assessment of the potential size and implications of this programme on the Eurosystem's balance sheet.in the long term.

# 2. HOW THE ECB'S MONETARY POLICY WORKS AGAINST PRICE STABILITY AND EU ENERGY TRANSITION

This section discusses the two-way relationship between the ECB's monetary policy and the green transition. On the one hand, a high interest rate environment negatively impacts the renewables sector disproportionally. Simultaneously, the delay in the renewables buildout leaves the European economy exposed to the negative effect of fossilflation, which may pose further risks for the conduct of price stability. This strongly intertwined relationship demonstrates the relevance of a successful energy transition for the ECB's mandate

# Box 1: The relevance of climate change for the ECB's primary and secondary mandate

The primary mandate of the ECB is to preserve price stability in the euro area by steering the short-term interest rate of the banking sector (ECB, 2021c).

In this context, the ECB has recognised already in 2018 that climate change would have an impact on price stability, given the increased frequency of extreme weather events and their implications in terms of climate-related physical and transition risks on households and corporates, as well as the greater volatility and uncertainty induced on relative prices and inflation expectations (Coeuré, 2018).

This understanding was only reinforced in the recent years. For instance, the ECB made climate change a central part of its previous Strategy review and its recently revised operational framework (ECB, 2021b, 2024a). The relevance of climate change for price stability has been reiterated throughout the year by various ECB officials(see for example Elderson, 2021; Schnabel, 2022), as well as by the academic literature (Bremus, Dany-Knedlik and Schlaak, 2020; Beirne *et al.*, 2024).

Moreover, Article 127 of the TFEU stipulates: "Without prejudice to the objective of price stability, the ESCB shall support the general economic policies in the Union with a view to contributing to the achievement of the objectives of the Union as laid down in Article 3 of the Treaty on European Union. This so-called "secondary mandate", or perhaps more accurately "supportive objective", has often been neglected by the ECB (van Tilburg and Simić, 2021; Van't Klooster and De Boer, 2021). However, in recent years, it has made a resurgence. For instance, the ECB Executive Board Member Frank Elderson stressed that observing the secondary objective was "a duty, not an option" (Elderson, 2021).

Whether and how the ECB can prioritise the fight against climate change under its secondary mandate crucially depends on the degree of prioritisation of this issue by the EU's political institutions, and the existence of clear policies in this regard (Ioannidis, Hlásková and Zilioli, 2021; van Tilburg and Simić, 2021).

With the adoption of the Fit for 55 package under the previous legislature, the European Union has set itself high ambitions to accelerate the transition towards a low carbon economy. For instance, the EU adopted a Renewable Energy Directive (RED) which sets a target to reach 42.5% of renewable energy production in the EU by 2030 (*EU Directive 2023/2413 on the Promotion of Renewable Energy*, 2023). The Energy Performance of Buildings Directive (EPBD) also set a binding target to increase the average energy performance of the national residential building stock by 16% by 2030 (*EU Directive 2024/1275 on the energy performance of buildings (EPBD recast)*, 2024), in order to reduce energy consumption. Moreover, the RepowerEU strategy supplemented the EU green Deal by reinforcing the EU's ambition to decrease its dependence on Russian gas.

Successful implementation of these policies will unambiguously require important investment needs, to the tune of  $\in$ 400 to 500 billion per years (Andersson *et al.*, 2024). It is also clear that the bulk of these investments will require a greater input from the banking sector.

In the last few years, climate change mitigation has thus been on the top of the EU's agenda, with the adoption of the EU Green Deal and Climate Law. Moreover, the EU's goal of strategic autonomy, against the backdrop of the war in Ukraine and wider geopolitical tensions, also loudly speak in favour of accelerating the shift in the EU's energy mix towards more domestically produced green energy. Supporting these objectives could thus be interpreted to fall under the ECB's secondary mandate of "contributing to the achievement of the objectives of the Union", as indicated in the Article 127. In the summer of 2021, energy prices started to rise globally, due to the geopolitical tensions running up to the outbreak of the war in Ukraine, in combination with the opening up of economies after Covid. Due to the rise of energy prices, inflation in the Eurozone rose to 10.2% at its peak in October 2022 (see Figure 1 below). The ECB's Governing Council responded to this by raising interest rates in July 2022 up to 4.5% – their highest point since the creation of the ECB – and by phasing out its asset purchase programme (see Figure 3 in section 3.1).

## Figure 1. Inflation contributors and total inflation in the Eurozone between 2021 and 2024



The ECB's strategy to combat energy-driven inflation faced criticism, as interest rates changes have been described by many as inadequate to address a supply shock, like in this case energy-driven inflation (Krahé and Heilmann, 2023; Van 'T Klooster and Weber, 2024).

Changes in interest rate apply to the overall economy. However, the impact of these higher rates is not homogenous across sectors, as some sectors are more sensitive to interest rates changes than others, due to their higher reliance on debt financing costs. Renewable energy, for instance, is a very capital-intensive sector (IEA, 2023; Kleintop, 2023). This is because the cost of generating renewable energy is predominantly falling on the high upfront investment costs to build and install the required infrastructure. The further input costs of wind or solar radiation are basically nil: the sun shines and the wind blows for free. The opposite is true for fossil fuel energy, whose infrastructure have already been largely built, but whose running costs are very dependent on its input cost, which are the cost of

commodities required to run oil and gas power plants. This difference in the cost structure has deep implications on the cost-sensitivity of each energy technologies to changes of interest rates.

## Empirical findings on the interest rate sensitivity of renewable projects

Multiple empirical studies confirm that renewable energies are more sensitive to changes in interest rates (Monnin, 2015; Schmidt *et al.*, 2019; Polzin *et al.*, 2021; Đukan *et al.*, 2023; Aguila and Wullweber, 2024). For instance, IEA finds that an interest rate hike from 3% to 7% would entail an increase of the Levelized cost of Electricity (LCOE) of more than 30% for offshore wind and solar PV (IEA, 2020). In contrast, the LCOE of gas-fired power would only increase by 4% (IEA, 2020).

Similarly, Voldsgaard et al. (2022) find that an increase of the cost of capital by 5 to 10% would lead to an increase of the cost of electricity from offshore wind, large-scale PV, and rooftop PV by 47%, 52-54% and 60% respectively. In contrast, the cost of capital hike would increase the cost of gas-fired electricity by only 8% (Voldsgaard, Egli and Pollitt, 2022). Anecdotally, a survey by the Netherlands Association for Renewable Energy found that a third of its members reported that an increased interest rate environment has negative impact on their financing options (NVDE, 2023a).

In a recent paper, ECB researchers found a different, but not contradicting conclusion. They found that green firms generally benefitted from interest rates 14 basis points lower than high-emission firms in the period 2018-2022. Moreover, high carbon firms were hit more aggressively (by 1bps) after a 25bps hike in the ECB key rate (Altavilla *et al.*, 2023). However, the study does not capture the structural disadvantage of renewable investments as aforementioned. A relatively smaller increase in costs of capital (as found by Altavilla and colleagues) may therefore still translate into larger total costs (i.e. LCOE) for renewables than fossil fuel projects.

At the same time, the ECB's own Survey on the Access to Finance of Enterprises (SAFE), finds that more than half of Eurozone companies indicate that high interest rates present very important obstacles to obtaining climate investments (Ferrando, Groß and Rariga, 2023).

Overall, the literature clearly establishes that interest rate hikes the ECB are increasing the cost of capital for renewable projects, which can appear to be in a strong contradiction with the ECB's mandate to support the above-mentioned economic policies of the EU. Unfortunately, this policy contradiction is likely to occur again, as geopolitical risks remain high. After its unprecedented peak in 2022, inflation has significantly come down, enabling the ECB to cut rates by 25 basis points in June 2024 (ECB, 2024d). However, the geopolitical risks factors that led to the fossilflation of 2021-2023 are still very much present. When trying to escape its dependency on Russian gas, the EU has led itself to diversify its energy supply to other external sources of gas, like Qatar and the US. But the underlying risk of energy price shocks remain generally similar, as exemplified by the disruptions in supply chains due to terrorist attacks in the Red Sea or technical difficulties in the Norwegian gas pipeline facilities (Buli, 2024; Gwyn Jones, 2024; Partington, 2024). In addition, the next US administration may be less favourable to EU trade conditions (Moens and Gijs, 2023; Milligan, 2024). Although such geopolitical risks have always been possible, their probability, intensity, and potential gravity, have arguably never been higher than today.

As we will see in the next section, the risk of a policy mismatch between monetary policy and the green transition may also undermine the pursuit of price stability by the ECB itself.

#### More renewable energy will benefit the price stability

An important aspect of the debate on whether central banks should support the green transition is linked with the question as to whether the energy transition will lead to overall higher or lower levels of inflation. The so-called "greenflation" defines the hypothesis where it would lead to higher inflation (Schnabel, 2022). However, the opposite hypothesis also exists, as illustrated by the unexpected decline in solar PV costs by 89% between 2010 and 2022 (IRENA, 2023). As Christine Lagarde stated: "investment in renewables, grids and storage will lead to lower and more stable energy prices" (Lagarde, 2024). If the President of the ECB is correct, then price stability would greatly benefit from a speedy energy transition. As the share of clean energy production increases, inflation overall will reduce. In this section, we will explore the empirical evidence underpinning this statement.

Preliminary evidence of this negative relationship between the share of renewable energy sources in the energy mix and inflation was found in Spain, where an increase of 1 GWh of electricity production with renewables or cogeneration reduced electricity prices by almost 2€/MWh during the period from 2005 until 2011 (Gelabert, Labandeira and Linares, 2011)<sup>1</sup>. Moreover, Bank of Spain researchers recently found that a larger share of renewable energy could lead to a 50% reduction of wholesale energy prices in Spain by 2030 (Quintana, 2024). Using data from all Eurozone countries, Bednář and associates find a similarly strong correlation between energy prices and inflation, especially since 2009. They find that for each percentage increase in the electricity contribution in the HICP correlates with an increase in the HICP by six percentage points, and that the share of renewable electricity consumption is negatively associated with inflation. (Bednář *et al.*, 2022). Analysis by the International Energy Agency arrives at a similar conclusion, finding that "without PV and wind capacity growth in 2021-2023, average wholesale electricity prices would be higher by about 3% in 2021, 8% in 2022 and 15% in 2023" (International Energy Agency, 2023).

One exception in the literature is a recent IMF working paper, in which the authors try to identify correlation between renewable energy production and lower inflation, using data from 75 countries over the 1973-2022 period (Millischer et al., 2024). However, the authors only find evidence of such correlation for the few EU countries with more than 85% of green renewable electricity production in their domestic supply mix, and not for other countries. As the authors explain, this is largely due to the specificities of the marginal energy pricing model that is being used in the EU energy market, where gas-powered electricity cost de facto determines the price of renewable energy<sup>2</sup>. This echoes the analysis of the EU's Joint Research Centre, which concludes that the benefit of renewable energy in terms of lower wholesale prices will take longer to materialise, due in part to limited storage capacity and system flexibility (Gasparella, Koolen and Zucker, 2023). In the EU's marginal price system, any linear increase in the share of renewables will not lead to an immediate and linear decrease to the whole wholesale energy price. There is instead a tipping point in the energy mix after which this correlation will fully come to effect.

All these findings tend to minimize the perceived risks of "greenflation". It is possible that higher prices of certain materials will increase, while the overall cost of production of green energy will also continue to decrease with technological advances and scaling up effects. Moreover, to reiterate, the cost of capital plays an important role in the output electricity cost, as we discussed in the previous section. Given that the cost of capital is partly a consequence of central bank's interest rate policies, there can be a feedback loop between central bank policies and energy inflation. Even if the greenflation hypothesis turned out to be true, it would still be counter-productive if central bank policies continue to increase the capital cost for renewables, since this would lead to higher production costs for green energy, ultimately translating into energy-driven inflation.

2 The wholesale electricity price in the EU is determined by the production cost of the energy technology which has to be switched on the last in order to meet new demand. Hence, once renewables and nuclear production is fully consumed, gas-powered plants are usually switched on to meet the EU demand. Given this, the price of gas most often determines the wholesale price of electricity, including when generated by renewables (Zakeri et al., 2023). Against this background it appears that raising interest rates has an unfortunate side-effect of negatively impacting the sectors – in this case renewables – that could help reducing the inflationary pressure in the future. From that point of view, although raising interest rates could be necessary on the short term, it could simultaneously be counterproductive policy for long-term price stability. In the next section we will review the recent policy discussion on the possibility of introducing a green interest rate into the ECB's monetary policy framework, in order to situate how this concept could be practically implemented, and the various constraints already identified.

16

Sustainable Finance Lab

# 3. BACKGROUND AND POLICY DISCUSSION

In this chapter, we will review the recent policy developments around longer term refinancing operations (LTROs) and the subsequent debate on the possibility to adopt green refinancing operations. This background will prove useful in understanding the policy constraints and concerns that the policy idea has generated so far, and which will guide our thinking when designing a Green TLTRO.

#### Short history of ECB's refinancing operations

Before diving into the policy discussion on the greening of central bank refinancing operations, we first briefly summarize the recent history of ECB refinancing operations. This may be a good basis for understanding how these operations work, and therefore how they could be "greened".

#### The early history

Before the great financial crisis of 2008-2009, there was an active interbank lending market and banks would usually lend to each other. The role for the central bank was to steer the interbank lending rate within a certain corridor defined by the key interest rates.

The provision of liquidity to banks was thus more limited and indirect. Banks had to compete to receive a portion of this limited liquidity by submitting bids, specifying both the amount of lending they want and the interest rate they are willing to pay. The central bank then ranked the bids by the level of interest rate offered and provided liquidity starting from the highest bids until the predetermined total amount was exhausted (Linzert, Nautz and Bindseil, 2006; Eisenschmidt, Hirsch and Linzert, 2009).

#### The post-crisis new normal

With the outbreak of the great financial crisis in 2008, the interbank market collapsed, leaving no choice to the central bank but to fill the gap by providing ample liquidity to the banking sector. This was done with the deployment of large-scale asset purchases programmes ("quantitative easing") and extension of already existing long-term lending operations (LTROs) beyond the usual 3 months maturity. To further support the banking sector during this turmoil, the central bank started offering loans against fixed rate and full allotment. In such a regime, banks do not have to compete anymore to receive liquidity: all bids submitted by banks are satisfied, up to a borrowing allowance. In 2014 the ECB introduced targeted LTROs (TLTROs) in order to strengthen the lending effects of the program (see Box 2 below).

The amount of these operations reached historically high levels, culminating at around 7 trillion of total outstanding liquidity injected in May 2021, including 2.2 trillion under TLTROs (see Figure 2 below).





Source: ECB (2024f, 2024c).

As the central bank injected more reserves into the banking sector, the central bank's deposit rate played a more powerful role than the refinancing rate, and the whole system shifted from a corridor regime to a so-called floor regime<sup>3</sup>. As Figure 3 below shows, the interbank lending rate (EONIA, in blue) used to fluctuate between the Deposit Rate (DFR, in orange) and the Main Refinancing rate of the

18

ECB (MRO) until 2012. After that, the EONIA rate closely follows the ECB's deposit facility rate.

## Figure 3. History of the of the ECB's monetary policy interest rates and instruments



In response to the Covid-19 pandemic crisis, the ECB crossed another Rubicon. The ECB lowered the interest rate under TLTRO III to as low as -1%, well below its -0.5% deposit rate, and with 4 years maturity (ECB, 2020). In this setting, banks borrowing from the ECB could actually make money: they pay less in interest rate than they receive from interest payments on their deposit accounts at the central bank. There was, however, one condition: that banks meet a "lending performance threshold" over a specific period – namely to increase their (non-mortgage) lending to households and non-financial corporations (ECB, 2019).

#### Box 2: Findings on the performance of (T)LTROs

The LTRO operations, the precursor to TLTROs, were suspected to lead to carry trades by recipient banks. This means that these banks could use the discounted central bank lending to purchase government bonds (or other financial instruments) with a higher yield to benefit from the interest rate difference. The literature confirmed that carry trades were indeed taking place (see for example Crosignani, Faria-e-Castro and Fonseca, 2015; Acharya, Pierret and Steffen, 2016).

19

A green interest rate for the eurozone

As a result, the ECB assigned firmer conditionalities to long-term lending in 2014 with the introduction of TLTRO-I – hence the update to the name of the program to *targeted* LTROs. The stated goal of this shift was strengthening the lending to the real economy and improving the monetary policy mechanism (ECB, 2014). The academic studies showed that this targeting was effective: there was little evidence of carry trade taking place, and the lending to the real economy improved in TLTRO programmes compared to earlier LTROs (de Haan, Holton and van den End, 2019; Laine, 2019; Benetton and Fantino, 2021; Castillo Lozoya, García-Escudero and Pérez Ortiz, 2022).

The TLTROs programme had positive effects across multiple other dimensions. For example, research has shown that banks that participated in the TLTRO-I program reduced the interest rates by 20 basis points and increased lending volume by 17% compared to non-participating banks (Benetton and Fantino, 2021). TLTRO-II was found to increase lending to the non-financial corporations by 20% (Laine, 2019). Vulnerable economies<sup>4</sup> of the Eurozone were found to have especially benefitted from the TLTRO programme, at least in 2016 and 2017 in terms of the amount of credit extended to the real economy (Afonso and Sousa-Leite, 2019). Moreover, certain studies find that TLTROs positively impacted macroeconomic variables such as GDP and inflation (Laine and Nelimarkka, 2023).

TLTROs are important examples of the way in which the ECB went beyond the conventional constraints of monetary policy. The innovative features of TLTROs inspired further creative suggestions as to repurpose refinancing operations to facilitate the green transition, as we will see in the following section.

#### **Emergence of Green TLTROs proposals**

At the end of 2019, the newly elected ECB President announced her intention to investigate the possible role of the ECB to support the fight against climate change, which by that time had become a pressing demand from a range of non-profit organisations and think tanks (Arnold, 2019).

Henceforth, the ECB embarked itself into a 18-month long strategy review process, which pave the ground for a wider public debate on how the ECB could address climate change. Inspired by Eric Lonergan<sup>5</sup>, the idea of designing TLTROs targeted specifically towards green lending emerged. "Green TLTROs" were defined as

5 Eric Lonergan was the first to describe "dual rates" as the possibility for the central bank to offer a lending interest rate lower than the deposit facility rate (Lonergan, 2016, 2019, 2020).

<sup>4</sup> In the study defined as: Cypress, Spain, Greece, Ireland, Italy, Portugal and Slovenia.

"refinancing operations where the interest rate that banks pay depends on their volume of lending that complies with the EU's Green Taxonomy" (Van't Klooster and Van Tilburg, 2020).

Hubert Kempf had also imagined the implementation of a "climate spread" on the interest rate applicable to the ECB's refinancing operations (Kempf, 2020). This interest rate differentiation would be based on a climate scoring of commercial banks, serving as either an incentive or a penalty for less environmentally friendly banks.<sup>6</sup>

The idea was also reviewed by the Network for Greening the Financial System (NGFS), and identified as one of the potentially most effective ways to mitigate climate change while also pointing out to the difficult operational feasibility (NGFS, 2021).

### Figure 4. The NGFS identified green refinancing operations as one of the potentially most impactful measures central banks could take



#### ECB's reaction to Green TLTROs

ECB President Lagarde initially welcomed the Green TLTROs proposal.<sup>7</sup> But in spite of her openness, the ECB did not end up introducing a differentiated green interest rate in the final conclusions of its strategic review. The Governing Council

6 We also acknowledge that Nikolas Z. Muller might have been the first who theorized the "green interest rate". However his approach differs from ours, as he defines the green interest rate as the equivalent of the natural interest rate which optimises consumption over the cycle depending on the expected path of pollution intensity of the economy. In his model, a green interest rate should therefore be higher than the normal rate, in order to decelerate consumption in a pollution-intensive economy (Muller, 2021).

7 At a hearing at the European Parliament on 28 September 2020, Christine Lagarde said: "you are asking me whether we would be prepared to consider this proposal as part of our strategy review, and the answer is yes." (European Parliament, 200).

21

deemed it "premature to concretely envisage targeted green operations at the current juncture." (Drudi et al., 2021) Several obstacles were identified, the first being the lack of reliable data on the green lending, since indeed, until very recently, commercial banks were not yet obliged to assess the Taxonomy-alignment of their loan books (Várhelyi, 2020). Another concern was ensuring that the policy would provide a level playing field for banks, as certain banks might be in a better position than others to achieve the green lending target (ECB, 2021a). Other concerns related to whether Green TLTROs do really fall within the mandate of the central bank, or whether it constitutes de facto a substitute for existing fiscal policies (Villeroy de Galhau, 2023).

Beyond these technical concerns, a more fundamental reason has been advanced to rule out the introduction of Green TLTROs. The ECB's strategy review led to the recognition of the importance of climate change for monetary policy under a "protective" or risk-based agenda, centred around the mitigation of climate-related risks and its implications for financial stability. By contrast, the introduction of differentiated rates is being seen as a more "proactive" or "allocative" approach where the central bank acts more directly in support of a reallocation of resources towards the low carbon transition. An proactive or allocative approach is often perceived as blurring the lines between monetary and fiscal policy (Boneva, Ferrucci and Mongelli, 2021; Kedward, Gabor and Ryan-Collins, 2022; Lane, 2024).

#### Resurgence of dual rates as a response to fossilflation

While the concept of dual rates was put aside for a while, it re-emerged in the discussion during the energy-led inflationary crisis following the war in Ukraine in 2022. While the ECB responded to this inflationary pressure by raising rates, critics of the ECB have argued that the "one size fits all" policy might not be sophisticated enough to address acute supply-shock stemming from imported fossil fuel energy prices. Indeed, while interest rate hikes may work when inflation is predominantly stemming from demand factors such as excessive wage growth, consumption or public budgets spending, the impact may be more limited when addressing supply shocks.

As we have seen in section 2, raising interest rates has an unfortunate side-effect of negatively impacting certain supply factors that could play a dis-inflationary role. From that point of view, although raising interest rates could be necessary on the short term, it could simultaneously be counterproductive policy for long-term price stability.

To overcome this trade-off, various voices from the banking sector and think tanks called once again on the ECB's governing council to adopt instead a "green dual rate" approach (Lonergan *et al.*, 2022; Monnet and Klooster, 2023; Van Doorslaer and Vermeiren, 2023). Green business leaders from France and the Netherlands

22

have also supported the idea at various occasions (NVDE, 2023b; Grandjean *et al.*, 2024), alerting about the impact of higher capital costs on their business model and on the levelized cost of energy (LCOE).

Going further, Van't Klooster and Weber have conceptualized a new inflation governance framework that would coordinate a broad range of stakeholders and policy areas including competition, fiscal, monetary policy, but also statistical authorities (Van 'T Klooster and Weber, 2024). Improving the governance of macrofinancial policy would go a long way to improve economic efficiency and crisis management in the European Monetary Union. A differentiated green interest rate as described in this report could be one of the instruments at the service of a more coordinated and optimal policy mix. A green interest rate would enable the central bank to act simultaneously on demand and supply factors. With a dual rate regime, the central bank could raise rates to reduce aggregate demand, while at the same time lowering the green interest rate to preserve specific supply side factors from undue tightening.

New research analysis also emerged showing that that the "additional lending from banks participating to the TLTRO III programme is mainly directed to more polluting sectors" (Colesanti-Senni, Pagliari and Van't Klooster, 2023). This advanced the case for a greener approach for the next period of monetary policy loosening. This argument was recognised by ECB Board member Frank Elderson: "whenever there is a monetary policy need in the future to reconsider targeted longer-term refinancing operations for banks, there are compelling reasons to seriously consider greening them" (Elderson, 2023).

Meanwhile, several central banks in Asia have recently introduced similar green lending schemes. In 2021, the People's Bank of China (PBOC) introduced a Carbon Emission Reduction Facility (CERF) enabling banks to borrow at a preferential rate of 1.75% for an amount corresponding to 60% of their volume of loans issued to support to the low-carbon transition (PBOC, 2021). The Bank of Japan followed suit in January 2022 by launching a new programme called "Transactions for Climate Response Financing Operations" under which banks can borrow at 0% from the Bank of Japan (Clarke, 2022). As of July 2024, Japanese banks were able to borrow 11,963 billion yen (€76 billion), following an almost constantly growing take-up (See Figure 5). Under both programmes, banks can self-identify the green loans eligible to the facility, but ex-post verifications are carried by the central bank.

In October 2021, the Hungarian central bank also launched a similar programme, targeted to a specific sector: green housing. Under the "Green Home Programme" initiative, the Magyar Nemzeti Bank provided 300 billion HUF (around €761 million) to credit institutions at a 0% interest rate under condition that these funds be used

for loans to households seeking to build or purchase new, homes with an energy performance certificate label higher than "BB" (80 kWh/m²/year) (NGFS, 2024).

#### Figure 5: Lending to Japanese banks under the Bank of Japan's Funds-Supplying Operations to Support Financing for Climate Change Responses



Source: Bank of Japan (2024).

#### Current state of discussion at the ECB

As we have seen so far, three major reasons prevented the ECB to decide on the implementation of Green TLTRO until now. First a definition of green lending, and availability of reliable data was missing. Second, a Green TLTRO framework should not interfere with the ECB's monetary policy and comply with the ECB's mandate. And third, the implementation of a dual rate strategy during a tightening period seemed complex and risky. We argue it is time for the ECB to re-evaluate the possibility of introducing a green interest rate.

Firstly, significant progress has been made in the provision of a reliable definition of green lending, and the availability of reliable data from the banking sector. In particular, progress in the adoption and early implementation of the EU Taxonomy represent a promising development which deserve to be taken into account, alongside other options.

Secondly, the ECB itself have now opened the path for reconsidering green dual rates when it announced in March 2024 its new operational framework, a plan for the ongoing reduction in excess reserves in the financial system (as discussed in section 3.1). Under this new framework, the ECB said it will continue to have structural refinancing operations alongside structural asset purchases. Importantly, for both these instruments, the ECB said it "will aim to incorporate climate change-

24

related considerations" (ECB, 2024a). Thus, the establishment of the new structural refinancing operations could provide a basis to deploy in the coming years some form of interest rate differentiation based on climate criteria. However, the ECB's new framework is only expected to come into place in the horizon of 2026-2027 (Schnabel, 2024).

Thirdly, the ECB expressed concerns against the introduction of a green interest rate policy under a restrictive monetary policy stance. For instance, ECB Board member Isabel Schnabel rejected the possibility of introducing dual rates, stating: "Green targeted lending operations, for example, could be an instrument worth considering in the future when policy needs to become expansionary again (...) But they are not an option for the immediate future given the current need for a restrictive monetary policy" (Schnabel, 2023).

Since June 2024, the ECB started to pivot its monetary policy strategy by cutting interest rates. If inflation continues to recede and the ECB continues to move towards a more accommodative stance, this will thus open the political space for differentiated rate.

In the following sections we will discuss the design choices for such a green differentiated rate program. Sections 4 to 6 will investigate key policy parameters of a green interest rate, with a view of addressing concerns that have been identified so far. Section 7 will summarise the policy trade-offs entailed in the implementation of a green interest rate, and provide provisional conclusions.

# A green interest rate for the eurozone

# 4. DEFINITION OF 'GREEN': USE OF THE EU TAXONOMY

As discussed, one of the main obstacles to the design of the green refinancing operations that the ECB foresaw originally was the lack of the good definition of a green asset and the lack of systematic data collection by banks (ECB, 2021a). Providing a definition of green lending is difficult for independent central banks, since they do not have the expertise nor the political legitimacy to specify which lending is green (Tamez, Weenink and Yoshinaga, 2024). Fortunately, the EU has recently adopted a framework that does just this: the EU Taxonomy of sustainable activities. In this section we will analyse to what extent can the EU Taxonomy be used for the purpose of implementing a green interest rate.

#### Introduction to the EU green Taxonomy

The EU Taxonomy was adopted in 2021 by the EU institutions in order to help investors, companies and policymakers to identify activities that are environmentally sustainable, aiming to facilitate the transition towards a greener economy and avoid greenwashing (European Commission, 2023).

As it stands, the EU Taxonomy covers economic activities including energy, manufacturing, transport, waste management, water, and buildings, among others. Importantly, it does not cover agriculture nor activities that are generally neutral in terms of carbon emission, like most services. The Taxonomy framework provides precise technical screening criteria, which are science-based thresholds and performance requirements that are tailored for each economic activity, and which must be met to qualify as environmentally sustainable. These criteria can be either positive or negative. For instance, a minimum energy-efficiency level for buildings represents a "Significant contribution criteria" while usage threshold of toxic material or water may constitute "Do No Significant Harm", or DNSH) criteria (European Commission, 2024).

#### Data availability and Green Asset Ratio

The EU Taxonomy applies to a broad range of actors, from corporations to financial institutions. Under Article 8 of the EU Taxonomy regulation, banks and insurers are required to disclose the Taxonomy-alignment of their EU-based loan books (European Union, 2020). Taxonomy-related disclosure requirements have been incorporated into banks pillar 3 disclosure requirements, with a harmonized reporting template<sup>8</sup>. Since 2024, all banks in the EU have to disclose annually their Taxonomy-alignment levels.

The most prominent KPI that banks have to disclose is the "green asset ratio" (GAR), i.e. the share of Taxonomy-aligned assets upon their total eligible assets. The GAR is accompanied by a GAR "Flow" which is meant to provide transparency on the institution's progress towards sustainable investment goals. The GAR Flow covers exclusively the share of new assets (originated or acquired during the reporting year) that are Taxonomy-aligned. However, for this first reporting cycle, many banks have not fully reported on their GAR Flow KPI<sup>9</sup>.

To calculate their GAR, banks must rely on the provision of climate related data by their customers. For loans to households, in particular mortgages, auto loans and renovation loans, the ESG relevant data can be relatively easily collected through Energy Performance Certificates (EPCs) which companies and individuals are increasingly obliged to obtain under EU and national law. While EPCs are sufficient to assess the alignment of loans with the Taxonomy's "significant contribution" criteria, banks often struggle to assess the compliance with the "Do Not Significant Harm" (DNSH) criteria<sup>10</sup>.

The situation is more complex for corporate loans, as data collection is contingent upon the implementation of the Corporate Sustainability Reporting Directive (CSRD). Under CSRD, corporates will be obliged to declare their own Taxonomyalignment, and such data will be exploited by banks in the calculation of their GAR. However, the CSRD implementation is gradual. It is only mandatory in 2025 for the largest firms, and in 2026 and 2027 for middle-size companies. Overall, 50,000 firms will ultimately have to comply with the CSRD, compared to 11,000 under the current NFRD rules. While the CSRD represents a huge step forward, it is not

9 For the reporting year 2023, only 70% of banks in our sample reported their GAR Flow. They justified this due to the lack of referential data for 2022. Banks which reported the GAR Flow often did not provide the nominal amount of new lending originated or acquired during the reporting period, which would be required for Green TLTRO.

10 For example, several banks interviewed mentioned difficulties with the DNSH technical screening criteria for building renovation activity which requires a "full flush volume of a maximum of 6 litres and a maximum average flush volume of 3,5 litres" (European Union, 2023).

<sup>8</sup> The Pillar 3 ESG reporting templates can be accessed here: <u>https://www.eba.europa.eu/activities/single-rulebook/regulatory-activities/transparency-and-pillar-3/implementing-technical-standards-its-prudential-disclosures-esg-risks-accordance-article-449a-crr</u>

foreseen to expand the CSRD scope to small companies<sup>11</sup>, although these entities can decide to voluntarily disclose their Taxonomy-alignment as well.

A key aspect of the GAR calculation formula is that certain assets are excluded from the numerator but not from the denominator. These are assets related to activities not classified by the Taxonomy (such as agriculture) or issued by SMEs not subject to reporting requirements under the NFRD or CSRD directive. By contrast, some assets (such as exposures to central banks, or cash) are excluded from both the numerator and the denominator. Therefore, there is an asymmetry in the way the ratio is calculated. This implies that, counter-intuitively, it is not possible for a bank to reach a green asset ratio of 100%. According to the EBA, the maximum GAR for the whole EU banking sector could be at most around 60% (European Banking Authority, 2021). This asymmetry is meant to ensure that the GAR level would not go down when more activities and/or counterparties will eventually be included in the EU Taxonomy (since this would increase the denominator).

#### Data sample from the banking sector

Since Taxonomy data is only available since early 2024, we have collected data for a number of banks in order to see by ourselves what data is now available. In section 8.2 we provide an overview of a sample of this newly available data. The most striking finding is that the GAR level of banks is rather low (2.23% on average) with a total amount of Taxonomy-aligned lending of €401 billion. We also observe high heterogeneity between the levels and quality of banks' green asset ratios. This is linked to their varying capacity to obtain the relevant data from their counterparts<sup>12</sup>, but also due to diverging practices in the assessment of the Taxonomy criteria<sup>13</sup>.

We also find that around 77% of Taxonomy-aligned lending ( $\leq$ 310 billion) are based on mortgages associated with energy efficient buildings. This dominance of real estate in the data may have important implications for the design of Green TLTRO, as we will discuss later. Unfortunately, outside of mortgages, renovations loans and car loans, the Taxonomy reporting templates for financial institutions do not require banks to provide granular information about the particular NACE sector codes or Taxonomy activities (although some banks do it on a voluntary basis).

12 For instance, collecting EPCs from mortgages is easier in countries where a national EPC database has been established.

13 Due to the late communication of a clarification notice by the EU Commission on 21 December 2023, some banks have decided to apply a strict approach following the new EC guidelines, while other have not implemented them for the 2023 reporting cycle.

<sup>11</sup> Due to the size criteria of CSRD, SPVs are most often outside of the CSRD scope. As SPVs are often used for renewable projects, this effectively excludes them from the GAR calculation. To remedy this issue, the EBA has proposed an additional voluntary-based KPI: the BTAR. The main difference with the GAR is that BTAR includes non-CSRD companies. However, banks do not yet report this indicator.

#### Implications for implementing a Green TLTRO programme

Overall, the first reporting cycle of the Green Asset Ratio reveals that the quality of these disclosures are still very much at the infancy stage, with high discrepancy in how these data are calculated by banks. Banks's Green Asset Ratios are particularly low. In this section, we will discuss to what extent these difficulties complicate the implementation of a green intertest rate by the ECB, and the possible ways to remedy these issues.

First of all, the fact that Green Asset Ratios are currently low would significantly limit the borrowing allowances of banks if Green TLTRO was implemented as of 2024. However, while banks' GAR are low, the total aggregate volume of Taxonomy-aligned lending was in 2023 already significant from a macroeconomic perspective (around €300-400 billion). And in any case, with the gradual implementation of CSRD and banks' ongoing efforts in improving their ESG data collection systems, it is almost certain that Taxonomy-alignment levels will increase rapidly in the coming years. The complexity and administrative cost to collect Taxonomy-relevant data will remain high in the early phase. However, the financial incentives provided by the green interest rate would partly compensate for it. The European Commission is expected to review the Taxonomy regulation in 2025<sup>14</sup>, which will be a window of opportunity to simplify the reporting rules as well as refine screening criteria. Complementary solutions could be envisaged as well. For example, the EBA has proposed the introduction of green labels for certain retail products, with potentially more flexible rules and criteria than the EU Taxonomy (European Banking Authority, 2023). Assets compliant with such EU label specifications could also be made eligible to Green TLTRO when such additional labelling system in place.

We also see a risk of data inconsistencies, or even greenwashing, especially since Taxonomy disclosures are not fully audited (yet). Certain banks could also be tempted to tweak their balance sheet composition (by reducing their non-Taxonomy-aligned lending, or selling these assets to other entities) in order to fictively improve their GAR. Banks could be tempted to do that especially near the end of the reference period, in order to artificially boost the figures that will eventually be formally reported in their annual reports. To cope with these two issues, Várhelyi (2020) suggested that independent third parties (e.g. audit companies) should be mandated by national central banks to verify the compliance of banks' reporting with the Taxonomy framework. This could be done in practice by carrying out sample-based verifications of individual assets (for example by verifying the energy performance certificates associated with mortgages and renovations loans). If a certain threshold of errors is detected, the auditing company could then proceed with even more granular verifications. The percentage of errors detected could be subtracted (including retroactively) from the ECB's preferential Green TLTRO rate. Significant and/or repeated misreporting should be subject to further sanctions, or even provoke the exclusion of banks from the programme.

A perhaps more complex issue relates to the conceptual nature of the EU Taxonomy, which takes a focus on economic activities that may be considered as "sustainable". For this reason not all activities are covered: for example agriculture and services are missing. Most SMEs are also not covered since they are not falling under the CSRD. As a result, banks specialized in financing SMEs and agriculture could be disadvantaged by a green interest rate based on the European Taxonomy.

At the same time, introducing a green interest rate could also precisely provoke a catalysing effect for banks, corporates and policy makers to overcome these issues. With the promise of having a lower green interest rate from the ECB, banks and their counterparts will have much stronger incentives to make the reporting efforts. Even companies not covered by CSRD (such as SPVs) might find it costworthy to produce voluntary-based reporting if they can obtain cheaper funding in return. The resulting improvement in the quality and quantity of Taxonomy data reports will ultimately help improve the management of climate-related risks by banking supervisors and central banks.

#### Broad or selective Taxonomy approach?

The EU Taxonomy is very broad, covering a wide variety of activities falling under six climate and environmental objectives. This comprehensiveness may lead to two issues.

First, there is a risk that as the use of the EU Taxonomy scales up in the coming years and the Green TLTRO-eligible lending expands as a result, the programme could become oversized and jeopardize the conduct of monetary policy. In section 5.3 we discuss the various ways in which the ECB could mitigate the issue.

Second, it is not immediately clear that all the EU Taxonomy activities equally need to be supported by a Green TLTRO programme in order to achieve the ECB's primary and secondary objectives. Indeed, from a monetary policy lens, not all the economic activities covered by the EU Taxonomy contribute directly to the reduction of fossil fuel dependency. In addition, not all activities may suffer from particularly high investment needs. For illustrative purposes, consider 3 concrete examples:

- Afforestation makes laudable contribution to reducing carbon emission, but does not immediately help reducing energy consumption or increase clean energy production, and thus contribute to shielding the Eurozone against fossilflation.
- Conversely, investments in hydropower plants, while valuable for the energy transition, have a relatively low investment gap (Pellerin-Carlin, Calipel and Bizien, 2024). Thus, decreasing the cost of these investments may not be absolutely needed.
- Another striking example is the ownership or acquisition of real estate properties with an already high energy performance, which dominates by far the amount of Taxonomy-aligned lending disclosed by banks thus far (see section 4.2). From a climate perspective however, the effect of incentivizing the purchase of already well-insulated homes through green mortgages with a green interest rate policy can be negligible. Indeed, while energy-efficient buildings consume less energy, and by extension, less fossil fuel, home purchase transactions do not per se reduce carbon emissions<sup>15</sup> it is the investment into energy-efficient renovations which does so.

For these reasons, the ECB could instead select a subset of the Taxonomy activities that would more effectively help fulfil its mandate. An obvious move in this direction would start by excluding the Taxonomy activity "Acquisition and ownership of real estate" from the Green TLTRO programme. Indeed, as pointed above, the green lending stimulus from a Green TLTRO would be better targeted towards funding for renovation as opposed to mortgage transactions for the purchase of already high energy efficient real estate.<sup>16</sup>

From a monetary policy standpoint, targeting Green TLTRO to specific sectors that have likely a more positive contribution to price stability, such as investments in renewables energy infrastructures, electrical grid interconnection, storage, and energy-efficient renovation, would likely address fossilflation risks more effectively. Moreover, a selective approach would also mechanically reduce the potential size of the Green TLTRO programme, which would conveniently limit potential interference with the overall monetary policy stance on the short-term. We will return to this question in section 5.2.

A green interest rate for the eurozone

15 Actually, the inclusion of real estate into Green TLTRO risks fuelling housing prices for well-insulated dwellings, and thus may represent a risk for financial stability. For this reason, the ECB already excluded loans for house purchase under TLTRO III.

16 In the case of "green mortgages" whose proceed cover both the purchase and the renovation of the property, the segment of the loan dedicated to the renovation should be eligible to Green TLTRO.

Going towards a selective approach would require more detailed Taxonomy reporting by banks – as NACE sectoral or Taxonomy activity breakdowns are not currently required in Pillar 3 disclosures. It would also require having a robust analysis of the inflationary contribution of each of the activities represented in the EU green Taxonomy. To our knowledge, such exercise has not yet been undertaken by the Eurosystem.<sup>17</sup> Yet, without such strong evidence-based analysis, it might be more complicated to justify the choice of excluding certain activities from the Taxonomy. The inclusion or exclusion of certain activities such as gas and nuclear energy, could also prove quite controversial and compromise the reputation of the ECB. To mitigate such risks, the formal (but advisory) involvement of other EU agencies with more sectorial expertise in this exercise could prove useful to provide political legitimacy for the ECB.

We will further summarize the trade-offs of choosing a broad or a selective Taxonomy approach, as well as their policy implications in the section 7. In the following section we will discuss the possible mechanisms underpinning the green interest rate policy.

# 5. STRUCTURE AND SIZE OF A GREEN TLTRO

The defining feature of the Green TLTRO is that is provides commercial banks with a lower interest rate offered by the central bank. There are various ways to determine how much lower should the green interest rate be (the "green spread"). Another key feature is the quantity of lending that should be extended to banks under a Green TLTRO policy. We will discuss in this section how the ECB could calibrate the optimal volume of the programme.

# Box 3: Understanding the Green TLTRO mechanism step by step

To illustrate concretely how the Green TLTRO programme would work, here is a simplified diagram describing the step-by-step process of a Green TLTRO operation, if implemented as soon as 2025 (See Figure 6 below). For the sake of simplicity, we assume the ECB would use a broad Taxonomy and Green TLTRO operations have a maturity of one year.

The fact that borrowing allowances are determined based on the lending of the previous reference period implies that banks receive Green TLTROs funding after having actually originated new green lending. This may seem counter-intuitive if one thinks of Green TLTRO as a "funding for lending scheme" in a literal sense, where banks would technically pre-require the TLTRO funding before they can start lending to their customers. However this does not reflect the functioning of credit and money creation in today's economic system (Jakab and Kumhof, 2015). Moreover, from a practical perspective, it would be impossible for banks and central banks to predict *ex ante* (at the date of the allotment) the amount of green lending that will be originated by banks in the future – and which should correspond to the borrowing allowance. As a consequence, the borrowing allowance and green interest rate incentive can only be calculated *ex-post*. However in practice banks will anticipate the opportunities offered by the Green TLTRO mechanism and optimize their lending behaviour accordingly.

#### Figure 6. Simplified description of the proposed mechanism underlying Green Long-Term Refinancing Operations with 1 year maturity



Source: authors.

#### Determining the green interest rate

The green interest rate mechanism could be determined by three components. These three mechanisms can be implemented on their own or combined:

- Banks would benefit from a "basic" green interest rate on their amount of Green TLTRO borrowing based on their previous green lending. This is the mechanism adopted by the Chinese and Japanese central banks.
- A "bonus" green rate would be applied to banks, but under condition that banks reach a predefined lending target. This was the mechanism of the ECB's TLTRO III.
- A penalty rate could be applied for banks with underwhelming green lending performance.

These three mechanisms offer different advantages and disadvantages, which we discuss below.

34

#### Basic green interest rate

Under a simple green interest rate rule, the preferential rate would be granted on the basis of the green lending issued by the bank. This green interest rate would in fact be expressed as a "spread" between the key ECB interest rate (for example, -100bps of the average MRO rate during the reference period<sup>18</sup>). This way, the green interest rate would automatically be adjusted upward or downward whenever the ECB's Governing Council decides to change its key interest rates.

The advantage of this option is that banks are guaranteed to receive a lower interest rate, and therefore they can confidently transmit this lower interest rate onto their new green loans. This contrasts with the mechanism of the TLTRO III, where the preferential rate was only granted to banks which reached their "lending performance threshold" – and thus correspond to the green bonus rate which we present in the next section. The risk of this option is that only banks that are certain to reach the target will tend to participate to the programme, or if they do, they will not immediately transmit the lower rate to their customers until they are certain to qualify for the bonus rate.

However, if implemented alone, a basic green interest rate would not necessarily encourage banks to "walk the next mile" and scale up their lending towards the levels needed to fill the EU's investments gaps. For this reason, the next section discusses the possibility to introduce a "green bonus rate" subject to a volumebased lending target.

#### Bonus green rate

The bonus interest rate is a similar mechanism than the ECB's TLTRO III programme. Under TLTRO III, banks would get an interest rate of 50 bps below the deposit facility rate if they managed to increase their outstanding lending to households (excluding loans for house purchase) and non-financial corporations by 1% (this threshold was reduced at 0% during the pandemic crisis). Banks that did not achieve the target had to pay the usual interest rate.

The key advantage of a target-based approach is that Green TLTRO would not just reduce the cost of lending, but also motivate banks to be more ambitious and aim for a higher volume of loan origination. This mechanism could thus be exploited to steer the whole banking sector towards filling climate investment gaps.

18 In the recent past, the ECB's deposit facility rate (DFR) was the main key interest rate. However as of September 2024, the ECB has narrowed the difference between the MRO and the DFR from 50 to 15bps, signalling the ECB's intention to re-establish a corridor regime where the MRO play once again an important role. For this reason, and for clarity, we use the MRO as referential for the green interest rate.

#### How to set the bonus rate lending targets

Defining lending targets under the EU Taxonomy approach is a delicate exercise, since the EU Taxonomy in itself does not provide transition pathways against which banks should be evaluated upon. In principle, the forthcoming transition plans foreseen in the CSRD could provide a robust tool for this, however the design of such plans are still at the infancy stage.

A more immediate and practical and generic approach would consist in using the Green Asset Ratio as a key performance metric<sup>19</sup>. For example, a bank with a current GAR of 4% would need to reach 6% the following year, while a bank with a GAR of 10% would need to reach 12%.

To define the "speed" of the GAR target, one could start by evaluating the GAR level that the whole banking sector should reach by 2030 or 2050 in order to help the economy meet its transition goals. From there, one can calculate retroactively the speed at which the GAR has to increase on average yearly to reach this level. For the purpose of illustration, let us assume that the banking sector should reach a GAR of 50% in 2050. In such a scenario, then banks would need to continuously increase their GAR by 2 percentage points every year. Hence, the ECB could set as a target for banks to increase their GAR by 2 percentage points every year.

Under a preliminary calculation, two percentage points of GAR increase would represent around €300-400 billion per year (see section 8.3). Although this volume is very tentative at this point, it is interesting to note that it is within the range of the various estimates of public and private annual climate investments need, which are comprised between 300-800 billion for the whole EU (Andersson *et al.*, 2024; Draghi, 2024).

To reiterate, banks can increase GAR by increasing the numerator (originating new lending or screening preexisting assets as green) but also by reducing the denominator i.e. reducing their holding of non-Taxonomy-aligned assets (which happens naturally as loans reach maturity, or when a bank sells off these assets). Thus, each bank has multiple leverages and possible strategies to reach the GAR target. In the short run, banks will likely mechanically increase their GAR as they roll-out their capacity to collect and analyse ESG data. Furthermore, as companies start complying with the CSRD, pre-existing loans may subsequently become Taxonomy-aligned.

The effects of different targets on bank lending can be hard to predict. If targets are too ambitious and difficult to achieve, it risks demotivating banks to even participate in a Green TLTRO. Moreover, uncertainty as to whether the bank will
qualify to the bonus rate at the end of the period may generate reluctancy to passover the lower rate to their customers, at least until the GAR target is within reach. Conversely, if the targets are too low and easy, banks simply receive a windfall profit and may not "walk the extra mile" as intended by this policy design choice. To mitigate this issue, the ECB could offer multiple-tier interest rates, with even more generous interest rate reduction in case the bank surpasses the basic lending target, or for the top-tier of banks that have the best green lending performance.

Overall, a combination of both mechanisms could be a powerful mix to ensure a broad participation of banks while also effectively pushing banks to scale up their green lending towards the volumes needed to achieve the transition.

#### Penalty rate

In addition to a (lower) green interest rate, a penalty rate could be established to penalize banks whose lending continues to fund the most polluting and carbonintensive projects.

In principle, such penalty rate could reinforce the incentives for banks to support the transition and avoid situations where banks benefitting from a lower green interest rate simultaneously continue to fund activities that are negative for the transition. Theoretically, the penalty rate could also play a reinforcing role during monetary tightening periods. Incidentally, such a penalty rate would reduce the cost of the policy on the Eurosystem's balance sheet (as discussed in section 8.5).

However, the operational feasibility of this option is rather complicated, given the absence of a well-developed "red" Taxonomy that negatively screens out activities that are not Paris-aligned. Moreover, the establishment of a penalty rate could discourage certain banks from taking part in the Green TLTRO programme. It would perhaps be more prudent to envisage introducing a penalty rate in a second stage, and only if such counter-productive lending behaviors are observed empirically.

#### Size of the "green spread"

The "green spread" should be big enough to motivate banks to develop new strategies to increase green lending, and to transmit in lower rates for green projects and consumers.

As the cost of funding reduces, banks will be able to offer lower rates for renewable projects. On top of that, the lower rate will reduce the costs of green projects and hence increase their profitability, which is likely to motivate or attract more project developers (or households willing to invest in home green energy), and thus help scale up green investments. Theoretically, the bigger the green spread, the stronger its impact will be. But how large should green spread be in order to materially help close the EU's green investment gap? Previous research showed that 100bps differentiation would help achieve 5 to 12% of the needed emission reductions EU goals (Altaghlibi, van Tilburg and Sanders, 2022). A more recent analysis found that a green rate that is 200bps lower would improve by 24% to 53% the profitability of most renewable technologies for solar and wind (onshore and offshore) for Germany, France, Italy, the Netherlands and Spain (Altaghlibi, van Tilburg and Bronstering, 2024). These results however widely differ depending on the future path of energy prices.

In his report for the European Commission published in September 2024, Mario Draghi estimates that a reduction of 250 bps of the cost of capital would be required to reach a level of private investment of around 4% of GDP – as required to bridge the EU's green investments (Draghi, 2024).

On the other hand, a too big green spread will increase the cost of the Green TLTRO on the balance sheet of the Eurozone central banks. We discuss and quantify this implication in section 8.5.

#### Borrowing allowance and size of Green TLTRO portfolio

Under the ECB's TLTROS III, lending was limited by a borrowing allowance set for each bank as 55% of their total of eligible lending (loans to non-financial corporations and households excluding mortgages). However, this rule was designed under a vastly accommodative monetary policy stance, where the issuance of abundant liquidity was an integral part of the ECB's strategy. But as we have seen in section 3.5, the ECB intends to move away from the abundant liquidity regime, and move towards a hybrid regime where the size of structural refinancing operations will be calibrated to correspond to the liquidity needs of the financial system (Schnabel, 2024). Various academics have generally estimated that such liquidity needs will be ranging from €1 trillion to €1.7 trillion (Gotti and Papadia, 2024). The ECB will therefore need to carefully calibrate the programme in order to avoid a situation where the Green TLTRO becomes bigger than its liquidity target. There are four ways in which the ECB can do that.

#### Reducing the scope of eligible lending

First, we should note that the decision to follow either a broad or selective approach as described in section 4.3 greatly determines the borrowing allowance of each bank. With a selective scope of eligible green lending the central bank would automatically limit the borrowing allowance, and therefore the potential size and growth of Green TLTROs. For example, excluding mortgages from the scope of Green TLTRO would reduce the size of the programme by four times according to our estimates (see section 8.3). Theoretically, the ECB could initially choose a "broad" eligible green lending scope, and decide later to switch to a more selective approach by narrowing down the scope of eligible green lending to a fewer number of economic activities that are the most relevant for price stability.

#### Reducing the borrowing allowance ratio

If the ECB does not choose a selective approach, then one option to limit the size of Green TLTRO is to reduce the borrowing allowance ratio for each bank. Instead of borrowing up to 100% of the green lending originated during the previous reference period, the exact borrowing allowance ratio could be calibrated over time. This adjustment could ensure that the total amount of is aligned with the central bank's monetary policy stance and liquidity forecasts. For example, if the ECB needs to reduce liquidity provision in the banking system, it could reduce the borrowing allowance for each bank, say to 80% instead of 100%, thus automatically reducing the volume of the whole programme. The People's Bank of China for example applies a borrowing allowance of 60% of green lending (PBoC, 2021).

#### Setting a hard limit on Green TLTRO size with competitive bidding.

The ECB could also pre-determine a maximum Green TLTRO benchmark allotment for the whole banking sector, thus creating a hard limit on the size of the programme. This approach would thus depart from the "full-allotment" policy that has become standard in the post financial crisis regime, although partial-allotment refinancing lines could perfectly coexist with (non-green) full-allotment structural refinancing operations. In this case, the ECB would pre-announce how much lending it intends to allocate to under Green TLTRO (the benchmark allotment). If the total bid amount exceeds the maximum benchmark allotment, banks would then have to compete to receive the liquidity.

In a first scenario, banks would have to compete by bidding on the interest rate through an auction, as was the case under LTROs until 2009. In this case the ECB would set a minimum green rate and banks would submit bids for green refinancing they are willing to take at smaller green spreads. The banks offering the highest rate would get their Green TLTRO allotment first until all accepted bids exhaust the maximum lending amount set by the ECB.

This option, however, would risk excluding certain banks completely from the operation. In addition, it might counteract the spirit of the Green TLTRO programme itself, as its goal is to reduce the interest rates in the first place. Decreasing the green spread by bidding in the auctions might then decrease the attractiveness of the programme for the participants.

#### Borrowing allowance limitation based on GAR

A more complex variant would work by reducing the relative borrowing allowance of each bank according to their past green performance. The ECB could apply a reduction of the borrowing allowance, based on for instance their share in the outstanding green lending of all banks participating in the auction. For example, a bank whose green lending stock represents 10% of the sum of bids from banks participating in the auction would only be able to receive up to 10% of the maximum lending amount. Thus, unlike the previous option, all banks' bids would be at least partly satisfied.

The (yet unconfirmed) assumption underlying this allocation is that banks with a high green lending level would be more likely to make an effective use of the liquidity to issue even more green lending in the future, given their greater experience and capacity to extend green lending.

However, if the ECB chooses to reduce the size of the programme, it is worth noting that this would eventually reduce the impact of the policy, as the lower interest rate would be offered against a smaller volume of new lending. The ECB could mitigate that by increasing the green spread. At the same time, as the amount of lending grows, it could also signify that the programme is reaching its goal to scale up green lending in the economy, and that the incentive may not be necessary anymore.

The magnitude of the 'green' spread and the determination of the programme size are one of the main considerations for the introduction of a Green TLTRO. In the following sections we discuss further design choices and overview some of their main trade-offs.

### 6. OTHER IMPLEMENTATION ASPECTS

Next to the determination of interest rates and volume size, ECB's lending programs are defined by several other parameters. In this section we review some of them. We first discuss which collateral rules the Green TLTRO program could follow. We then examine the maturity of the loans (the period after which the loan has to be repaid), the frequency of auctions, and the programme duration. We finish the section with considerations on communication of the launch of such programme.

#### **Collateral policy**

All refinancing operations from central banks usually require that banks pledge appropriate assets as collateral in exchange from central banks liquidity. The ECB's collateral framework determines which assets are eligible to be pledged as collateral for these operations (ECB, 2023). Several proposals have discussed limiting eligibility of assets issued by fossil fuel companies and promoting the securities by green ones (Oustry *et al.*, 2020; Dafermos *et al.*, 2021; Reclaim Finance, 2023). This would improve the financing conditions of green companies and penalize the polluting ones. In 2024, the ECB considered the possibility of implementing pool-limitations on intensive carbon assets but concluded in July 2024 that such option was too complicated (ECB, 2024b).

Theoretically it is possible to constrain the provision of Green TLTRO liquidity against the provision of green collateral (such as green bonds). However, the pool of green collateral is likely too small and regulatory green standards are not available for all types of assets, especially securitized assets (EBA *et al.*, 2013). As a result, such an approach would drastically reduce the potential borrowing available for banks. Overall, maintaining the standard collateral policy rules might add to the simplicity of the program, as well as alleviate concerns around the disruptions of monetary policy transmission. In other words, while Green TLTRO liquidity would only be provided under condition that banks issue green lending it is not necessary that banks pledge "green collateral" in order take part in Green TLTRO. However, if or when new green rules are eventually introduced in the Eurosystem collateral framework, extending these rules to apply to Green TLTRO might be reasonable, given the climate orientation of both policies.

#### Maturity

Central bank refinancing operations have typically a shorter maturity, given that these policies are designed to influence the short-term interbank market rate. While certain TLTROs operations had a maturity of up to 4 years, the ECB's current refinancing operations (LTROs) have a maximum duration of only 3 months (ECB, 2024e).

If the maturity of Green TLTROs is shorter than the maturity of the eligible green lending, this reduces the attractiveness of the Green TLTRO, since the reduction in the cost of capital will only apply for a shorter period after which the green lending has to be refinanced by the bank.

For instance, the maturity of loans for energy efficient renovations is typically around 10 years, or more. If the Green TLTRO maturity is only one year, this means that the underlying cost of funding for the renovation loan is not reduced during the 9 remaining years<sup>20</sup>.

Thus, the longer the maturity, the stronger the incentive for banks to issue green loans. Ideally, the maturity of the Green TLTRO loans should correspond approximately to the maturity of the green lending that such programme is designed to incentivize.

However, providing long maturities also implies that the central bank commits to maintaining the programme for a longer time span, which can raise difficulties for the conduct of its monetary policy. In particular, this makes the policy less easily reversible, although the ECB can always propose early repayments options for banks. Moreover, if the central bank offers important long-term liquidity provision, it may result in a weaker transmission of its policy on the short-term repo market rate. In addition, central bank funding operations with a residual maturity shorter

20 Although, as bank loans are usually amortizable with gradually diminishing principal repayments, the major part of the interest rate cost comes in the first few years. In contrast, the principal of a bond is usually repaid in a lump sum at maturity date.

A green interest rate for the eurozone

43

than one year do not qualify for the net stable funding ratio (NSFR) under prudential rules and are therefore much less attractive for banks (EBA, 2015). A mid-way solution can be to have a short maturity combined with the possibility of rolling over. For example, the Bank of Japan's climate transaction programme offers loans with a one-year maturity, which can be rolled-over (Bank of Japan, 2024). This programme enables banks to receive cheaper funding for longer, while preserving the ability for the central bank to phase out the programme more quickly if it would ever be needed.

#### **Allotment Frequency**

The next questions relate to how often the ECB should enable banks to borrow more money from the Green TLTRO programme.

In general, banks might prefer more frequent allotments, as this allows them greater ability and flexibility to obtain liquidity they need. For this reason, the ECB's Main Refinancing Operations are made on a weekly basis. However, a green LTRO is not designed to provide short term liquidity to banks. The objective would be lowering the cost of green lending and increasing lending volumes on the medium term.

Moreover, experience from previous TLTRO operations has shown that such a programme can be complex to administer – both for central and commercial banks. To minimize the complexity, cost and administrative burden of Green TLTRO, the frequency of the allotments could be reduced to an annual basis (instead of quarterly under TLTRO III). Furthermore, for the sake of simplicity, the operation schedule should be aligned with the Taxonomy report and Pillar III disclosure deadlines (usually end-June) in order to streamline the work of the banks' reporting teams.

#### **Programme duration and communication**

To stimulate new lending, a Green TLTRO program should run for a substantial period to send a strong signal to banks and allow them to integrate the incentives into their business strategies. If the ECB starts this programme in 2025, a duration of five years could be appropriate. As the programme would conclude in 2030, this would also enable a meaningful evaluation of the program's impact in supporting the EU's green transition toward the 2030 targets.

Although the ECB could theoretically extend the program beyond 2030, or even make it open-ended, having a set end date could incentivize investors and banks to expedite project management planning before the end of 2030 to maximize the benefits of the Green TLTRO programme while it is available. In any case, the sooner the ECB announces this programme, the sooner banks can start incorporating this in their business strategy and at the operational level. The ECB should announce this programme at least one year earlier than the first Green TLTRO transaction, in order to allow every bank to speed up the process of identification of eligible loans in their loan books, in order to optimize their initial borrowing allowance.

Sustainable Finance Lab

## 7. DESIGN CHOICES AND POLICY TRADE-OFFS

In the previous chapter we have discussed in details the various policy design choices that the ECB should take into account when designing a green refinancing programme. This section intends to clarify the multitude of options for each parameter and their combinations. To this effect, we formulate four general guiding principles relevant for the success of the programme, before assessing each policy option according to these principles.

#### **General Principles**

Taking into consideration the ECB mandate and policy discussion, we identify four general principles. These are the conditions that a Green TLTRO should meet and hence the criteria that any design could be assessed upon.

#### Monetary policy compatibility

A differentiated rate approach should be operational regardless of whether the central bank is conducting a restrictive or expansionary policy stance. To ensure this is the case, the Green TLTRO should not interfere or contradict the overall monetary policy strategy or blur its communication. Moreover, the ECB should preserve sufficient flexibility and discretion to adjust the parameters of the programme in order to adapt to the evolving economic outlook and monetary policy stance.

#### **Broad participation**

Since the goal of this programme is to reduce the cost of investments into clean energy production and energy efficiency in the whole Eurozone, it should be open to all banks from all size across all Eurozone member states and that a sufficiently large number of them do participate. This way, the policy will transmit its effect to all corners of the Eurozone economy. It is thus crucial that the program offers a level playing field to all banks.

#### Stimulate new lending

The goal of this programme is to stimulate banks to develop robust strategies to effectively increase their financing to the green transition in the coming years. Hence, the green spread should be large enough to significantly reduce the cost of borrowing for corporates and households, and the green lending scope should be adequately defined to areas that have financing needs.

#### **Operational feasibility**

Last but not least, the programme's rulebook and administrative burden should be made as simple as possible for banks, but also for the central banks that will administer it.

#### Comparative assessment of design choices

While all options are technically feasible, some poses specific advantages and downsides regarding the general principles we outlined in the introduction to this section. Table 1 below summarizes our findings.

### Table 1. Comparison of Green TLTRO design features againstfour defined principles

Darker green options indicate stronger preference. Lighter greens options indicate potential concerns.

Principles Green lending	Monetary policy compatibility definition	Broad participation	Green lending stimulus	Operational feasibility
Broad Taxonomy	A broad Taxonomy scope could stimulate activities unrelated to fossilflation and thus not directly contribute to price stability (e.g. adaptation, water, circularity). In the case of real estate, it could even create financial stability risks.	All banks are obliged to use the EU Taxonomy, so they should be capable to onboard the programme and develop their own strategies.	Incentivizes all types of Taxonomy- aligned lending, including those that may not be so relevant for the energy transition (e.g. adaptation, biodiversity, water).	All banks and regulators have already started using the EU Taxonomy, as are a growing number of companies. Reporting frame- works are in place.

A green interest rate for the eurozone

Principles	Monetary policy compatibility	Broad participation	Green lending stimulus	Operational feasibility
Selective Taxonomy	Can be effectively targeted to stimulate dis- inflationary sectors.	Depending on which sectors are chosen, certain banks could be disfavoured due to their lending portfolios or business models.	The effect of green interest can be better targeted to the most critically needed activities, and therefore stimulate these stronger.	A selective approach would require additional ad-hoc reporting from existing Pillar 3 templates. Central banks might also be reluctant to make sectoral allocation decisions.
Green interest	rate mechanism			
Basic green interest rate (only)	The level of green spread can be pegged to the ECB key interest rate in order to adjust automatically with the overall monetary policy stance.	Creates a positive incentive for all banks to participate regardless of their expected lending growth.	The effect might be more limited since banks can benefit from a windfall effect (they receive a lower rate for loans they would have issued anyway).	Easy to implement
Bonus green interest rate (only)	The level of green spread can be pegged to the ECB key interest rate in order to adjust automatically with the overall monetary policy stance	Could limit the participation to banks that are already certain to reach the target.	Lending targets create a strong incentive for banks to aim for higher quantity of green loans, however, the green interest rate might not transmit auto-matically to final consumers until the bank is certain to have reach the target.	Relatively more complex to imple- ment than the basic green interest rate.
Combination of lending- based and bonus rate	The level of green spread can be pegged to the ECB key interest rate in order to adjust automatically with the overall monetary policy stance	All banks are certain to benefit from the incentives of Green TLTRO	Quantitative targets create a strong incentive for banks to develop more ambitious, volume-based strategies	Relatively more complex to implement than using only one mechanism.

47

Principles	Monetary policy compatibility	Broad participation	Green lending stimulus	Operational feasibility
Penalty rate	The penalty rate would play a marginal role, unlikely to interfere with the monetary policy stance.	Will likely discourage banks to participate, especially if a penalty rate is established in the early phase.	Will likely reinforce incentives for banks to focus in green lending	Complex to implement, especially due to the lack of Taxonomy for dirty assets.
Limiting size of	the programme			
Borrowing allowance with full- allotment	Risk that Green TLTRO overshoots its intended size (especially if a broad Taxonomy scope is chosen), although the central bank can always limit the % of borrowing allowance to reduce the size of Green TLTRO liquidity in line with monetary policy stance.	Enables all banks to participate.	Borrowing allowance rule provides guarantee that banks can access Green TLTRO if they issue more green lending. However, the impact would be reduced in case the ECB reduces the borrowing allowance ratio over time.	Simple, follows the current practice of full-allotment operations.
Hard limit with auction based on interest rate bids	Hard limit on the size of Green TLTRO minimizes risk of injecting excess liquidity, maintaining? monetary policy stance.	The competitive auction risks excluding banks whose bids are not selected due to their low legacy green lending issuance.	The competitive auction will result in a higher green interest rate.	This used to be the standard design of refinancing operations.
Hard limit based on relative share of GAR level among participating banks	Hard limit on the size of Green TLTRO minimizes risk of injecting excess liquidity, preserving the monetary policy stance.	The auctioning scheme will limit the borrowing allowance of each bank, however all banks will be able to participate to some degree.	Hard limit may result in suboptimal amount of Green TLTRO green lending to accommodate investment needs	Relatively more complex to implement

A green interest rate for the eurozone

Principles	Monetary policy compatibility	Broad participation	Green lending stimulus	Operational feasibility				
Collateral policy								
Standard collateral rules	In line with current monetary policy framework.	Enables all banks to participate.	No particular difference	In line with the current design of refinancing operations.				
Green collateral requirement	Shortage of green assets in the current collateral pool of banks will reduce the provision of Green TLTRO liquidity.	Risk to drastically reduce how much eligible green lending banks can pledge from Green TLTRO due to shortage of green assets.	This will possibly limit the borrowing allowance of banks that have more eligible lending than green collateral eligible assets. Although in the long run, this could reinforce incentives for banks to issue more green collateral-eligible assets.	Requires making important changes to the existing Eurosystem collateral framework, which would delay the programme implementation.				
Maturity								
Short (<1 year)	Aligned with current monetary policy instrument geared at steering short-term interest rate.	Less attractive for banks since the Green TLTRO funding would not qualify for Net Stable Funding.	The Green TLTRO incentive is less significant if the maturity is shorter, (unless if the Green TLTRO can be rolled over).	Shorter maturity implies more frequency (see point below).				
Long (>1 year)		Long maturity operations are attractive for banks to acquire stable funding.	The Green TLTRO incentive is stronger with a longer maturity	Longer maturity requires less frequent operations, which reduces the administrative burden for banks and central banks				

Principles	Monetary policy compatibility	Broad participation	Green lending stimulus	Operational feasibility
inequency				
Quarterly	Enables ECB to adjust the policy parameters more often, for example to match to the liquidity analysis for monetary policy.	No particular difference	No particular difference	Creates more complexity and administrative burden for banks and central banks
Yearly	Induces more rigidity in the ECB's liquidity management.	No particular difference	No particular difference	Less complexity and administrative burden for banks and central banks

As the table shows, the list of policy parameters identified could enable a multitude of design choices and combinations. We note the following key points: First and foremost, the choice of the green lending scope is a critical one, as it tends to have important consequences in deciding on other parameters. Typically, a broader Taxonomy approach could lead to a larger size of the programme, so it could necessitate additional action to limit the size of it. On the other hand, choosing a selective approach would, by design, limit the size of the Green TLTRO and thus avoid having to resort to hard measures to contain the size of the programme. A middle way between the two would be to decide whether to start small with a selective approach while possibly extending to other activities in the medium run, or whether to take a broader approach in the first phase, while being ready to narrow down the scope of the programme on the longer run.

When choosing the interest rate mechanism, a combination of both a basic green interest rate and a target-based bonus rate reduction could provide the most optimal and complementary mix: while a lending-based green interest rate cannot steer banks towards ambitious volumes of green lending, a target-based bonus rate does. Vice-versa, a green bonus rate only can refrain banks from participating and have a lower transmission to the real economy, whereas a basic lending rate does not have that disadvantage.

When it comes to managing the size of the Green TLTRO programme, a scheme based on standard borrowing allowance could be the most effective, except in the case where a broad Taxonomy scope is chosen. As more sectors are by definition covered by the broad than the selective approach, this could risk making the programme too big in the medium run. To be on the safe side, the ECB would have to be prepared to adjust downward the borrowing allowance ratio (or other discussed policy parameters) in case the Green TLTRO size grows bigger than intended. If the ECB assesses that keeping the rules unchanged during the whole programme duration is preferable, then setting a "hard limit" would be the second best.

In general, there is a potential trade-off between stimulating green lending on the longer term and the conduct of monetary policy. In particular, more generous conditions in terms of longer maturity and programme duration would clearly maximise the potential impact of this programme. On the other hand, this could also create constraints for the conduct of monetary policy, as Green TLTRO could interfere with the contracting policy stance on the short term. However, as already indicated, there are effective instruments for central banks to reduce this risk.

As already discussed, constraining the Green TLTRO to green collateral rules would be both complex and counterproductive, at least on the very immediate future. However, a Green TLTRO programme could still be coupled with other type of climate-related adjustments to the collateral framework if such changes were decided in the future by the ECB.

We also note that having only yearly Green TLTRO auctions (compared to quarterly TLTROs) could be an important way to reduce the complexity and administrative cost of the programme, assuming the ECB's liquidity management can still operate in this context.

## 8. PRELIMINARY PROGRAMME SIMULATIONS

To get an impression of the quantitative implications of a Green TLTRO at the current stage of Taxonomy implementation, we built a unique dataset Taxonomyalignment of the biggest European banks, based on their disclosure reports for 2023 – the very first reporting cycle for the GAR. This database allows us to run some basic simulations for illustrating the magnitude of the impact of a Green TLTRO on the overall lending of banks, and the impact on the profit and losses of the central banks of the Eurosystem. This section presents the database and its implications.

#### The dataset

For the purpose of this analysis, we have compiled a dataset of Taxonomyalignment disclosure of more than 73 banks established in the Eurozone, among the 110 banks supervised by the ECB-SSM. We excluded banks not headquartered in EU20 Eurozone countries, and most importantly, Eurozone banks who have not yet fulfilled their obligation to disclose their Taxonomy reports by the cut-off date of our sample (July 2024)<sup>21</sup>. Our sample thus covers 80.7% of banking assets in the Eurozone. We analysed the annual report (or separate pillar 3 or sustainability reports) of these banks in order to extract selected data points on their Taxonomyalignment.

Based on this sample, we calculate the weighted-average GAR (based on turnover) for the Eurozone banking sector at around 2.53%, the average GAR is 2.23% and the median GAR at 1.42%. 11% of banks in our sample have a GAR smaller than 0.1% and around a quarter (24.5%; 18 banks) have a GAR comprised between 0.1 and 1%. The banks with the highest GAR level in our sample are the 10 banks have a GAR higher than 5%, among which only three have a GAR higher than 10% (see

Figure in Annex). The complete list of banks' green asset ratio is available in the Annex.

In total, the nominal outstanding volume of Taxonomy-aligned banking asset is €398 billion. Unfortunately, a breakdown of Taxonomy-aligned asset per NACE sector or Taxonomy activity is not available under the current reporting requirements. This limitation implies, for instance, that it is not possible to identify how much Taxonomy-aligned lending has been extended towards renewable energy generation.

The vast majority (77,4%, 310 billion) of Taxonomy-aligned lending corresponds to mortgages whose underlying real estate asset has a high energy efficiency. Meanwhile, loans to financial corporations represent nearly 6% of assets, and 14% for non-financial corporations subject to the NFRD. However, renovation loans represent a minuscule amount with nearly €446 million of assets (0.11% - so small it does not appear in the chart below), at an equal level than loans to local government. Loans for vehicles are slightly more visible with €9 billion (2.3%). Loans for renewable energy infrastructure are included in financial or non-financial corporations.

### Figure 7. Type of Taxonomy-aligned assets held by a sample of Eurozone banks.



Source: authors' own work, based on annual reports of 73 EU banks.

#### Policy parameters

For this exercise, we decide to opt for a broad Taxonomy approach. This choice is primarily motivated by the fact that data on Taxonomy-alignment of banking assets have become available since early 2024. To run a similar exercise with a

53

"selective" approach, a more detailed breakdown of bank Taxonomy disclosures per NACE sector or Taxonomy activity would be necessary. We then use the following parameters:

Policy parameter	Design choice
Green interest rate	100 bps below the MRO <sup>22</sup>
Green bonus rate	50 bps reduction in addition to the green interest rate
Eligible green lending	Broad Taxonomy
Collateral policy	Standard
Borrowing allowance	100% of new Taxonomy-aligned lending at the end of
	the previous year
Lending target	GAR +2pp
Reference period	Previous reporting year
Maturity	One year, no rollover
Programme duration	Five years

#### **Potential size of Green TLTRO**

As discussed in section 5.3, the size of the programme will be an important consideration for the overall design of Green TLTRO and its compatibility with the ECB's monetary policy strategy.

Under a simple borrowing allowance rule, the size of the programme will be directly proportional to the amount of eligible Taxonomy-aligned lending issued by banks during the previous period. Our database enables us to have some preliminary idea about the potential size of a Green TLTRO programme. We find that in 2023 the banks in our sample held €401 billion of Taxonomy-aligned assets<sup>23</sup>. However, this amount corresponds to a stock of loans, and not to the flow of new lending originated in 2023. Given that 2023 was the very first reporting cycle, we have no reference data for 2022, and therefore data on green lending flows are not systematically available yet via Pillar 3 disclosure reports.

As a proxy, we use our dataset to simulate how much new green lending banks should theoretically extend in order to increase their GAR to the target of 2 percentages points in 2024. All things equal, we find that banks in our dataset would have to originate €317 billion of new eligible lending (including real estate) for the next reference period in order to reach the GAR target. Extrapolating beyond our sample, this would result in a rough estimate of €378 billion for the whole Euro area banks. For each individual bank, this represents a lending target

23 Our sample covers 80.7% of Eurozone banking assets. A simple extrapolation of our sample would result in a theoretical total volume of Taxonomy-aligned assets of 479 billion euros for the entire Euro area banks.

of around €4 billion on average per bank (median: €1.8 billion) (see Figure in Annex).

As already pointed out, in practice banks can also increase their GAR by reducing their holding of non-Taxonomy-aligned assets (which happens naturally as loans reach maturity, or when a bank sells off these assets). In the short run, it is also very likely that the GAR will increase mechanically as banks roll-out their capacity to collect and analyse ESG data, and as companies start complying with the CSRD. Under a selective Taxonomy approach, the simple exclusion of loans to households collateralized by real estate assets from the Green TLTRO scope would shrink the programme to around €90-108 billion.

For comparison, since 1999 the ECB lent on average  $\in$  674 billion to the banking sector throughout open market operations (MRO, LTROs, TLTROs). TLTROs alone peaked at  $\in$  2200 billion in 2021. The Green TLTRO therefore could be relatively small compared to its predecessors.

#### Incentives from the green interest rate: case studies

Given the incompleteness of data on the GAR Flow, we are unable to simulate the (retroactive) effect of Green TLTRO on our sample of banks. For illustrative purposes, we compare the outcome of the abovementioned interest rate rules on a sample of fictive and simplified banks with a balance sheet of €100 billion, with various growth levels of eligible lending (see Table 2 below).

We then calculate the interest rate that would be applicable for each bank depending on whether this bank qualifies for the target-based bonus interest rate. We also show the incentive for the bank against the counterfactual case where such bank would have borrowed the same amount at the MRO rate. Under this simulation, we take the simplistic assumption that all banks achieve their GAR growth by issuing more green lending – and not throughout changes in their loan book composition. We do this over 5 years to ensure the simulation shows the long run effect.

Although these are simplified examples, these allow us to identify patterns in the way a Green TLTRO programme would incentivize banks.

First, we notice that all banks have an incentive to take part in the Green TLTRO, albeit at different scale and speed. Accordingly, bank C, the Laggard, with the highest yearly growth rate is the one that will benefit from the highest interest rate deduction within the period despite its lower starting GAR level. However, this was not the case for the whole period: during year 1, this bank was the one benefiting less from the incentive, given its low starting point.

#### Table 2. Cumulative interest rate gain over 5 years

€ Millions	Starting level of GAR	GAR growth per year	Year 1	Year 2	Year 3	Year 4	Year 5	Total
<b>Bank A</b> Normal bank, high growth	2.5%	2.5%	56	56	58	59	61	290
<b>Bank B</b> Normal bank, slow growth	2.5%	1.0%	44	18	18	18	18	115
<b>Bank C</b> Laggard, catching up	0.5%	3.0%	11	90	94	97	101	393
<b>Bank D</b> Green pioneer, slow growth	5.0%	1.0%	88	18	18	18	18	159
<b>Bank E</b> Pioneer, steady growth	5.0%	2.5%	113	56	58	59	61	346

Inversely, the "green pioneer" banks both win the most in the first year given their higher GAR growth in year 0. However, the incentive quickly reduces for the bank with a slow growth rate of 1%, just like bank B. This illustrates that over time "green banks" do not have a superior advantage under the proposed rules. Only banks with a constantly high GAR growth level benefit the most from Green TLTRO.

#### The cost for the Eurosystem

One concern regarding this proposal is the cost of the proposal for a Green TLTRO on the central bank balance sheet. This problem is particularly acute at the moment, given the historically big losses made by Eurosystem central banks since 2022, due to the higher interest rate paid on the large amount of excess reserves in the post-pandemic period (Arnold, 2024). Depending on calculations, it is estimated that Eurosystem central banks made losses to the tune of €55 billion to €160 billion between 2022-2024, and will continue in the coming years until around 2027. (Belhocine, Vir Bhatia and Frie, 2023; De Grauwe and Ji, 2024; Fitch Ratings, 2024). The mechanism behind these losses is the following. When the central bank lends money to banks, this mechanically creates additional reserves that commercial banks unavoidably deposit back to the central bank, at the deposit facility. Although it was not always historically the case (Monnet, Humann and Mitchener, 2024), the central bank usually remunerates commercial banks deposits at the central bank. For this reason, lending operations can bear a cost for the central bank in case the central bank lends money at a lower interest rate than the money it pays on the deposit facility, as would be the case under Green TLTRO.

However, the losses made from a Green TLTRO programme are likely to be notably smaller than these, given the smaller size of the Green TLTRO compared to the addition of TLTROs and the APP. We ran a simple simulation to approximate the potential cost of a Green TLTRO programme on the Eurosystem's balance sheet.

As in the previous section, we first assume that all banks increase their GAR by 2 percentage points, which would (by simplification) entitle them to a total borrowing allowance of €317 billion. As we discussed above, this number gives a generous estimate for the early phase of Green TLTRO, but is within the range of the possible size of the programme if banks were to successfully increase their GAR by 2 percentage points per year in the next few years.

Such injection of central bank reserves via Green TLTRO would increase deposits held by banks at the ECB's deposit facility, resulting in around €11 billion of interest rates payments from the Eurosystem to banks<sup>24.</sup> Meanwhile, as all banks would qualify for the bonus green interest rate they would all qualify to a green spread reduction of 150bps below the MRO, thus having to pay an interest rate of 2.15% (MRO 3.65% – 150bps), resulting in an income of €6.8 billion for the ECB. The difference between the two operations constitutes 4.2 billion in losses for the Eurosystem. This cost rises to 4.7 billion when considering the opportunity cost due to the loss of income that would have been generated by the central bank if it lends money at the MRO rate instead of the Green TLTRO rate<sup>25.</sup> If we extrapolate this amount to the whole Euro Area banking sector (and not just the 73 banks in our sample), the losses would amount to €5.7 billion. If the Green TLTRO maturity was longer than one year, this number would have to be multiplied accordingly.

Although this simulation is taking overly simplistic (but generous) assumptions, they allow us to conclude that such annual losses would be substantially lower than the losses made by the Eurosystem in recent years, and should therefore not

24 For the sake of simplicity, we ignore the effects of the non-remuneration of minimum reserves requirements and of the heterogenous distribution of bank reserves across banks and member states.

25 This is also a conservative assumption. In the absence of a Green TLTRO scheme, banks could also seek funding on financial markets instead of the ECB's MRO.

be a major concern or obstacle to considering the introduction of a Green TLTRO policy. Moreover, technical solutions such as tiered reserves mechanisms are available to reduce the magnitude of this cost. For example, the ECB could expand further the scope of its unremunerated reserves by revising upward the level of minimum reserves requirements, as proposed by Paul de Grauwe and others (De Grauwe and Ji, 2024). This would effectively reduce the base amount to which the ECB is remunerating banks through its DFR, reducing its outflows and thus total losses.

In theory, a simple solution to completely avoid any losses would be to set the lowest possible green interest rate at the same level as the ECB's deposit facility rate. In such a situation, both rates neutralize each other. However, this solution would abruptly limit the incentive offered to banks, especially given that since September 2024, the spread between the MRO and the DFR has been reduced to 15bps (instead of 50bps before). In such a setup, the ECB would need to compensate such tiny green spread by making other policy parameters more attractive, for example by extend further the maturity of Green TLTROs. Another more radical solution would be to apply a "penalty rate" (higher than the MRO) against banks that continue to extend lending towards the dirtiest companies in parallel to their issuance of green lending (as discussed in section 5.1). In doing so, the ECB would generate more income from certain banks, which could offset – at least partly – the losses incurred by other banks. However, as discussed already such solution is technically difficult and could be counterproductive.

## 9. CONCLUSION

Since 2020, there has been a rich discussion on the possibility of introducing a green interest rate in the Eurosystem's policy framework. Since then, the economic rationale for moving in this direction seem even stronger in the context of high and volatile energy prices, growing geopolitical and geo-economic tensions, and a recurring risk of fossilflation. As outlined in the recent report by Mario Draghi, the EU's ambitions to increase its strategic autonomy and accelerate the decarbonation of the energy requires a massive increase in public and private investments in the transition. At this stage, it is not clear whether future development of the Capital Markets Union agenda will be sufficient to unlock new funding in the short term. Against this background the deployment of more proactive green monetary policy tools in support of the EU's strategic priorities and the implementation of the EU Green Deal could be justified under both the primary and secondary mandate of the ECB.

The technical obstacles and justifications advanced previously by the ECB for not introducing Green TLTROs do no longer seem valid. With the development of Taxonomy-related mandatory disclosure for banks and corporations, it is now possible to evaluate the volumes of green lending within a bank's balance sheet. Although the EU Taxonomy reporting framework could be improved, the EU Taxonomy can already greatly facilitate the implementation of a Green TLTRO programme in the near future. Moreover, a commitment to introduce Green TLTRO in the future would likely produce a catalysing effect in the banking and corporate sectors to accelerate the ongoing efforts to comply with the Taxonomy and CSRD requirements and thus increase their green asset ratio more rapidly. The design of a green interest rate policy remains a delicate exercise. Fortunately, policymakers have a variety of design choices at their disposal. A critically important parameter will be to choose between a broad or selective approach to the EU Taxonomy. Typically, a broader Taxonomy approach could be simpler to implement. However, it could also lead the programme to overshoot in size, which would necessitate careful and more complex calibration of other parameters. A selective approach would, by design, limit the size of the Green TLTRO and could more precisely target sectors relevant for fighting the fossilflation. On the other hand, the selective approach might require more "allocational" decision making that some central banks may not be willing to undertake in isolation from political authorities. Either way, it would be conceptually possible to start small with a selective approach while possibly extending to other activities in the medium run, or conversely to take a broader approach in the first phase, while being ready to narrow down the scope of the programme on the longer run. On the very short-run, adopting strict green collateral requirements would be both complex and counterproductive.

When it comes to the calculation method of the green interest rate, a basic interest rate on all green lending might offer simplicity, but a bonus green rate could increase banks' origination of new lending, as evidenced under previous TLTROs. Overall, a combination of both mechanisms could be a powerful mix, where banks with persistent GAR growth emerge as clear winners. A penalty rate could theoretically further reinforce the transmission mechanism, but would probably be best introduced in a later stage, for example once a 'red' Taxonomy becomes available in the EU.

In general, there is a potential trade-off between stimulating green lending on the longer term and the conduct of monetary policy. In particular, more generous conditions in terms of longer maturity and programme duration would clearly maximise the potential impact of this programme. On the other hand, this could also create constraints for the conduct of monetary policy, as Green TLTRO could interfere with the contracting policy stance on the short term.

Using a unique dataset of Taxonomy-alignment of Euro area commercial banks, we find that the existing stock of Taxonomy-aligned assets in the Euro area banking sector was comprised between €400-500 billion at end 2023. On the basis that banks would have to increase their Green Asset Ratios by 2 pp per year in order to qualify for the green bonus rate, the size of the Green TLTRO would reach an annual size comprised between €317-378 billion (€4 billion on average for each bank). This order of magnitude is in line with the range of investment gaps being discussed in the literature. Moreover, this falls within the possible envelope of liquidity that the ECB envisages to supply to banks under its future operational framework.

Overall, it appears that the risk that a Green TLTRO overgrows to the point of interfering with the ECB's monetary policy would be manageable. The ECB can easily limit the programme's growth by reducing the scope of eligible lending (i.e. excluding certain Taxonomy activities) or reducing the borrowing allowance ratio (i.e. banks can receive less liquidity than their past green lending). The most extreme, but administratively simple policy would be putting a hard cap on the size of the Green TLTRO programme and have banks compete to receive liquidity from this limited amount. An important design choice will be to decide whether to start small with a selective approach while possibly extending to other activities in the medium run, or whether to take a broader approach in the first place, while being ready to narrow down the scope of the programme later on.

Based on the literature, we find that a green spread of 100 to 200bps would already have tangible effects in stimulating renewable investments. However further research is needed to calibrate the green spread from a monetary policy perspective, and to evaluate the possible spill-over effects.

Due to negative cash-flow, resulting from the difference between the newly introduced green interest rate and the ECB's deposit facility rate, a Green TLTRO would bear a cost on the Eurosystem's balance sheet. Using conservative assumptions, we estimate that an interest rate reduction of 150bps on  $\leq$ 317 billion of Green TLTRO lending would imply a yearly cost of around  $\leq$ 5 billion on the Eurosystem. This is a much smaller amount than the recent losses made by the Eurosystem, and should therefore not constitute a primary concern when considering the adoption of such policy.

The ECB's forthcoming operational framework offers a roadmap for introducing climate-friendly structural refinancing operations in the coming years. This represents progress since the Strategy Review in 2020-2021, when the ECB rejected the idea. However, these greener structural operations might come too little and be of relatively small size to address the issues of fossilflation and supporting the EU political objectives related to climate change. As the ECB's monetary policy stance is expected to become more accommodative in the coming months, the ECB should seize this momentum to introduce a separate interest rate for green activities. Such initiative would also come at a timely moment to support the delivery of the EU's climate targets for 2030.

# ANNEX I: FIGURES FROM THE DATASET

### Figure 8. Green Asset Ratio levels of 73 EU banks (based on data reported by banks)

de Volksbank N.V.				10 000
Münchener				11 75%
			10.10%	11.73%
ING Groen		6.0	10.10%	
		E 0.4%	1078	
AIR Group pla		5.94%		
Kutyobank S A		5.94%		
Kutxabarik, S.A.		5.70%		
La Ballque Fostale		5.50%		
RCI Banque SA		5.20%		
Contederation		5.07%		
Crelan SA; Crelan NV		4.70%		
Banco de Sabadell, S.		4.41%		
Coöperatieve		4.12%		
BPCE S.A.		3.98%		
ABANCA Corporación	3	3.29%		
Crédit Agricole S.A.	2.9	96%		
Piraeus Financial	2.9	91%		
Banco Comercial	2.9	90%		
Intesa Sanpaolo S.p.	2.65	1%		
SFIL S.A.	2.60	%		
Eurobank Ergasias	2.50%	%		
Bank of Ireland Group	2.43%	6		
Banco Santander, S.A.	2.40%			
Caixa Geral de	2 30%			
COMMERZBANK	2.28%			
AL PHA SERVICES	2.10%			
Banco RPM S n A	2.10%			
Unionio Bonoo S A	2.02%			
Nordoo Book Abo	2.02%			
	2.00%			
AS SED PAIK	2.00%			
National Bank of	1.80%			
Banco de Crédito	1.73%			
BANCA MEDIOLANUM	1.61%			
Bankinter, S.A.	1.54%			
BPER Banca S.p.A.	1.50%			
Ibercaja Banco, S.A.	1.50%			
Société Générale S.A.	1.42%			
Landesbank Hessen-	1.40%			
Deutsche Bank AG	1.30%			
Bpifrance	1.24%			
Banque et Caisse d'	1.17%			
UniCredit S.p.A.	1.16%			
BAWAG Group AG	1.15%			
Banca Popolare di	1.14%			
Credito Emiliano	1.07%			
Iccrea Banca S.p.A	1.06%			
CaixaBank SA	1.05%			
Nova Liublianska	0.99%			
Cassa Centrale	0.98%			
DekaBank Doutecho	0.90%			
BNP Paribas S A	0.77%			
Erete Croup Bank AC	0.77%			
BANCA MONTE DEL	0.53%			
BANCA WONTE DET	0.50%			
DDV/A Dense Bilbas	0.50%			
DovA - Danco Dilbao	0.49%			
Rameisen bank	0.42%			
Landesbank Baden-	0.33%			
Barclays Bank Ireland	0.30%			
Deutsche	0.24%			
Norddeutsche	0.23%			
KBC Group NV	0.20%			
DZ BANK AG	0.15%			
HSBC Continental	0.13%			
Bayerische	0.12%			
Volkswagen Bank	0.12%			
Belfius Banque SA;	0.07%			
Hellenic Bank Public	0.02%			
OTP Luxemboura S à	0.00%			
Deutsche Apotheker-	0.00%			
Kuntarahoitus Ovi	0.00%			
Bank of Cyprus	0.00%			
EINECOBANK BANCA	0.00%			
	0.00%			
AS LEV Group	0.00%			
0.0	0%	5.00%	10.00%	15.0
		GAR turnover		

#### Figure 9. Volume of new lending required per bank to reach a 2 pp target under Green TLTRO programme (own calculations)



# **BIBLIOGRAPHY**

Acharya, V., Pierret, D. and Steffen, S. (2016) *Lender of Last Resort* versus Buyer of Last Resort – The Impact of the European Central Bank Actions on the Bank-Sovereign Nexus. Available at: https://www.econstor.eu/bitstream/10419/129113/1/848984226.pdf.

Afonso, A. and Sousa-Leite, J. (2019) *The transmission of unconventional monetary policy to bank credit supply: evidence from the TLTRO*. Banco de Portugal. Available at: https://www.bportugal.pt/sites/default/files/anexos/papers/wp201901\_0.pdf.

Aguila, N. and Wullweber, J. (2024) 'Greener and cheaper: green monetary policy in the era of inflation and high interest rates', *Eurasian Economic Review*, 14(1), pp. 39–60. Available at: https://doi.org/10.1007/s40822-024-00266-y.

Altaghlibi, M., van Tilburg, R. and Bronstering, G. (2024) *The impact of a Green LTRO on the energy transition*. ABN Amro. Available at: https://www.abnamro.com/research/en/our-research/esg-economist-the-impactof-a-green-ltro-on-the-energy-transition.

Altaghlibi, M., van Tilburg, R. and Sanders, M. (2022) *How much of a help is a green central banker*? Working Paper. Sustainable Finance Lab. Available at: https://sustainablefinancelab.nl/en/paper/how-much-of-a-help-is-a-green-central-banker/ (Accessed: 10 April 2024).

Altavilla, C., Boucinha, M., Pagano, M. and Polo, A. (2023) 'Climate Risk, Bank Lending and Monetary Policy'.

Andersson, M., Nerlich, C., Pasqua, C. and Rusinova, D. (2024) 'Massive investment needs to meet EU green and digital targets'. Available at: https://www.ecb.europa.eu/press/fie/box/html/ecb.fiebox202406\_01.en.html (Accessed: 18 June 2024).

Arnold, M. (2019) 'Christine Lagarde wants key role for climate change in ECB review', *Financial Times*, 27 November. Available at: https://www.ft.com/content/61ef385a-1129-11ea-a225-db2f231cfeae (Accessed: 23 August 2024).

Arnold, M. (2024) *ECB makes first loss since 2004 due to higher interest costs*. Available at: https://www.ft.com/content/735ec122-15f6-4ccc-9ec2-c2d93a8f95bf (Accessed: 28 August 2024).

Bank of Japan (2024) *Outline of Transactions for Climate Response Financing Operations, Bank of Japan.* Available at: https://www.boj.or.jp/en/mopo/measures/mkt\_ope/ope\_x/opetori22.htm (Accessed: 27 August 2024).

Bednář, O., Čečrdlová, A., Kadeřábková, B. and Řežábek, P. (2022) 'Energy Prices Impact on Inflationary Spiral', *Energies*, 15(9), p. 3443. Available at: https://doi.org/10.3390/en15093443.

Beirne, J., Dafermos, Y., Kriwoluzky, A., Renzhi, N., Volz, U. and Wittich, J. (2024) 'Weather-related disasters and inflation in the euro area', *Journal of Banking & Finance*, 169, p. 107298. Available at: https://doi.org/10.1016/j.jbankfin.2024.107298.

Belhocine, N., Vir Bhatia, A. and Frie, J. (2023) 'Raising Rates with a Large Balance Sheet: The Eurosystem's Net Income and its Fiscal Implications', *IMF Working Papers*, 2023(145), p. 1. Available at: https://doi.org/10.5089/9798400244643.001.

Benetton, M. and Fantino, D. (2021) 'Targeted monetary policy and bank lending behavior', *Journal of Financial Economics*, 142(1), pp. 404–429. Available at: https://doi.org/10.1016/j.jfineco.2021.05.002.

Boneva, L., Ferrucci, G. and Mongelli, F.P. (2021) 'To Be or Not to Be "Green": How Can Monetary Policy React to Climate Change?', *ECB Occasional Papers* [Preprint]. Available at: https://doi.org/10.2139/ssrn.3971287.

Bremus, F., Dany-Knedlik, G. and Schlaak, T. (2020) 'Price Stability and Climate Risks: Sensible Measures for the European Central Bank', *DIW Weekly Report* [Preprint]. Available at: https://doi.org/10.18723/DIW\_DWR:2020-14-1. Buli, N. (2024) 'Norway outage pushes European gas prices to highest this year', *Reuters*, 3 June. Available at:

https://www.reuters.com/markets/commodities/norway-gas-export-plummetssleipner-outage-shuts-nyhamna-plant-2024-06-03/ (Accessed: 5 September 2024).

Castillo Lozoya, C., García-Escudero, E.E. and Pérez Ortiz, L. (2022) *The Effect of Tltro lii on Spanish Credit Institutions' Balance Sheets*. Banco de España. Available at: https://www.bde.es/f/webbde/SES/Secciones/Publicaciones/InformesBoletinesRevi stas/ArticulosAnaliticos/22/T2/Files/be2202-art09e.pdf.

Clarke, D. (2022) 'BoJ green loans scheme gets underway', *Green Central Banking*, 20 January. Available at: iea (Accessed: 21 May 2024).

Coeuré, B. (2018) 'Monetary policy and climate change'. *Scaling up Green Finance: The Role of Central Banks*, Berlin, 8 November. Available at: https://www.ecb.europa.eu/press/key/date/2018/html/ecb.sp181108.en.html (Accessed: 5 June 2024).

Colesanti-Senni, C., Pagliari, M.S. and Van't Klooster, J. (2023) 'The Co2 Content of the Tltro III Scheme and its Greening', *De Nederlandsche Bank Working Papers* [Preprint]. Available at: https://doi.org/10.2139/ssrn.4613820.

Crosignani, M., Faria-e-Castro, M. and Fonseca, L. (2015) *Central Bank Interventions, Demand for Collateral, and Sovereign Borrowing Costs.* Available at: https://www.bportugal.pt/sites/default/files/anexos/papers/wp201509.pdf.

Dafermos, Y., Gabor, D., Nikolaidi, M., Pawloff, A. and van Lerven, F. (2021) *Greening the Eurosystem Collateral Framework: How to Decarbonisethe ECB's Monetary Policy.* New Economics Foundation. Available at: https://neweconomics.org/uploads/files/Collateral-Framework.pdf.

De Grauwe, P. and Ji, Y. (2024) 'How to conduct monetary policies. The ECB in the past, present and future', *Journal of International Money and Finance*, 143, p. 103048. Available at: https://doi.org/10.1016/j.jimonfin.2024.103048.

Draghi, M. (2024) *The future of European competitiveness*. European Commission. Available at: https://commission.europa.eu/document/download/97e481fd-2dc3-412d-be4c-f152a8232961\_en (Accessed: 9 September 2024).

Drudi, F., Moench, E., Holthausen, C., Weber, P.-F., Ferrucci, G., Setzer, R., Nino, V.D., Barbiero, F., Faccia, D., Breitenfellner, A., Faiella, I., Farkas, M., Bun, M., Fornari, F., Ciccarelli, M., Darracq Paries, M., Giovannini, A., Papadopoulou, N., Parker, M., Petroulakis, F., Hebbink, G., Piloiu, A., Hlásková, S.J., Ploj, G., Popov, A.A., Kapp, D., Lopez-Garcia, P., Lozej, M., Lydon, R., Manninen, O., Roos, M., Manzanares, A., Santabárbara, D., Schotten, G., Mikkonen, K., Sotomayor, B., Stracca, L., Mongelli, F.P., Montes-Galdón, C., Müller, G., Nerlich, C., Osiewicz, M., Cruz, L.V., Page, A., Avgousti, A., Diez-Caballero, A., Aubrechtova, J., Mendoza, A.F., Téllez, M.D., Grüning, P., Garcia-Sanchez, P., Isgro, L., Bua, G., Kashama, M.K., Queiroz, R., Rognone, L., Röhe, O., McInerney, N., Meinerding, C., Tsalaporta, P., <!>Van Den End, J.W., Theofilakou, A., Weth, M., Osorno-Torres, B. and Ouvrard, J.-F. (2021) 'Climate Change and Monetary Policy in the Euro Area', *European Central Bank Occasional Papers* [Preprint]. Available at: https://doi.org/10.2139/ssrn.3928292.

Đukan, M., Gumber, A., Egli, F. and Steffen, B. (2023) 'The role of policies in reducing the cost of capital for offshore wind', *iScience*, 26(6), p. 106945. Available at: https://doi.org/10.1016/j.isci.2023.106945.

EBA (2015) EBA Report On Net Stable Funding Requirements Under Article 510 Of The CRR. EBA. Available at: https://www.eba.europa.eu/sites/default/files/documents/10180/983359/bb66d52d-4eb1-43c9-8924-5054d46c125f/EBA-Op-2015-22%20NSFR%20Report.pdf (Accessed: 9 September 2024).

EBA, ESMA, ECB, and EIOPA (2013) 'Joint ESAs-ECB Statement on disclosures on climate change for structured finance products'.

ECB (2019) ECB announces details of new targeted longer-term refinancing operations (TLTRO III), European Central Bank. Available at: https://www.ecb.europa.eu/press/pr/date/2019/html/ecb.pr190606~d1b6e3247d.en.h tml (Accessed: 10 September 2024).

ECB (2020) ECB recalibrates targeted lending operations to further support real economy. Available at: https://www.ecb.europa.eu/press/pr/date/2020/html/ecb.pr200430~fa46f38486.en. html (Accessed: 10 September 2024).

ECB (2021a) *Climate change and monetary policy in the euro area.* European Central Bank. Available at: https://www.ecb.europa.eu/pub/pdf/scpops/ecb.op271~36775d43c8.en.pdf.

ECB (2021b) *Climate change and the strategy review, European Central Bank.* Available at: https://www.ecb.europa.eu/home/search/review/html/climatechange.en.html (Accessed: 9 September 2024). ECB (2021c) *Decisions, statements & accounts*. Available at: https://www.ecb.europa.eu/mopo/decisions/html/index.en.html (Accessed: 5 September 2024).

ECB (2023) *Collateral*. Available at: https://www.ecb.europa.eu/mopo/coll/html/index.en.html (Accessed: 6 September 2024).

ECB (2024a) Changes to the operational framework for implementing monetary policy. Available at: https://www.ecb.europa.eu/press/pr/date/2024/html/ecb.pr240313~807e240020.en. html (Accessed: 9 April 2024).

ECB (2024b) Decisions taken by the Governing Council of the ECB (in addition to decisions setting interest rates). Available at: https://www.ecb.europa.eu/press/govcdec/otherdec/2024/html/ecb.gc240719~dde1 2c2121.en.html (Accessed: 6 September 2024).

ECB (2024c) Lending to euro area credit institutions related to MPOs denominated in euro - Eurosystem, Euro area (changing composition), Weekly. Available at: https://data.ecb.europa.eu/data/datasets/ILM/ILM.W.U2.C.A050000.U2.EUR (Accessed: 18 September 2024).

ECB (2024d) *Monetary policy decisions*. Available at: https://www.ecb.europa.eu/press/pr/date/2024/html/ecb.mp240606~2148ecdb3c.e n.html (Accessed: 10 September 2024).

ECB (2024e) *Open market operations, European Central Bank.* Available at: https://www.ecb.europa.eu/mopo/implement/omo/html/index.en.html (Accessed: 6 September 2024).

ECB (2024f) Securities held for monetary policy purposes - Eurosystem, Euro area (changing composition), Weekly. Available at: https://data.ecb.europa.eu/data/datasets/ILM/ILM.W.U2.C.A070100.U2.EUR (Accessed: 18 September 2024).

ECB, E.C. (2014) ECB announces monetary policy measures to enhance the functioning of the monetary policy transmission mechanism, European Central Bank. Available at: https://www.ecb.europa.eu/press/pr/date/2014/html/pr140605\_2.en.html (Accessed: 2 September 2024).

68

Eisenschmidt, J., Hirsch, A. and Linzert, T. (2009) 'Bidding Behaviour in the ECB's Main Refinancing Operations during the Financial Crisis', *SSRN Electronic Journal* [Preprint]. Available at: https://doi.org/10.2139/ssrn.1396702.

Elderson, F. (2021) 'Greening monetary policy', *European Central Bank*, 13 February. Available at: https://www.ecb.europa.eu/press/blog/date/2021/html/ecb.blog210213~7e26af8606. en.html (Accessed: 21 May 2024).

Elderson, F. (2023) 'Monetary policy in the climate and nature crises: preserving a "Stabilitätskultur". Bertelsmann Stiftung, Berlin, 22 November. Available at: https://www.ecb.europa.eu/press/key/date/2023/html/ecb.sp231122~e12db02da3.en. html (Accessed: 18 June 2024).

*EU Directive 2023/2413 on the Promotion of Renewable Energy* (2023). Available at: https://eur-lex.europa.eu/eli/dir/2023/2413/oj (Accessed: 23 August 2024). *EU Directive 2024/1275 on the energy performance of buildings (EPBD recast)* (2024). Available at: http://data.europa.eu/eli/dir/2024/1275/oj/eng (Accessed: 23 August 2024).

European Banking Authority (2023) *EBA report on green loans and mortgages*. EBA/REP/2023/38. European Banking Authority. Available at: https://eba.europa.eu/sites/default/files/2023-12/e7bcc22e-7fc2-4ca9-b50db6e922f99513/EBA%20report%20on%20green%20loans%20and%20mortgages\_0.p df (Accessed: 19 June 2024).

European Central Bank (2019) *ECB announces details of new targeted longer-term refinancing operations (TLTRO III), European Central Bank.* Available at: https://www.ecb.europa.eu/press/pr/date/2019/html/ecb.pr190606~d1b6e3247d.en.h tml (Accessed: 25 June 2024).

European Commission (2023) *EU taxonomy for sustainable activities*. Available at: https://finance.ec.europa.eu/sustainable-finance/tools-and-standards/eutaxonomy-sustainable-activities\_en (Accessed: 6 September 2024).

European Commission (2024) *EU Taxonomy Navigator, European Commission*. Available at: https://ec.europa.eu/sustainable-finance-taxonomy/ (Accessed: 3 September 2024).

European Union (2020) *REGULATION (EU) 2020/852*. Available at: https://eurlex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:32020R0852 (Accessed: 10 September 2024). European Union (2023) Annex to the Commission Delegated Regulation (eu) 2020/852. Brussels. Available at: https://finance.ec.europa.eu/system/files/2023-06/taxonomy-regulation-delegated-act-2022-environmental-annex-2\_en\_0.pdf (Accessed: 18 September 2024).

Ferrando, A., Groß, J. and Rariga, J. (2023) 'Climate change and euro area firms' green investment and financing – results from the SAFE'. Available at: https://www.ecb.europa.eu/press/economicbulletin/focus/2023/html/ecb.ebbox202306\_05%7Ef5ec994b9e.en.html (Accessed: 21 June 2024).

Fitch Ratings (2024) *Eurozone Central Banks Face Further Losses*. Fitch Ratings. Available at: https://www.fitchratings.com/research/sovereigns/eurozone-centralbanks-face-further-losses-09-09-2024 (Accessed: 10 September 2024).

Gasparella, A., Koolen, D. and Zucker, A. (2023) *The Merit Order and Price-Setting Dynamics in European Electricity Markets, JRC Publications Repository*. Available at: https://publications.jrc.ec.europa.eu/repository/handle/JRC134300 (Accessed: 9 September 2024).

Gelabert, L., Labandeira, X. and Linares, P. (2011) 'An ex-post analysis of the effect of renewables and cogeneration on Spanish electricity prices', *Energy Economics*, 33, pp. S59–S65. Available at: https://doi.org/10.1016/j.eneco.2011.07.027.

Gotti, G. and Papadia, F. (2024) *The European Central Bank's operational framework and what it is missing*. Working paper. Bruegel. Available at: https://www.bruegel.org/working-paper/european-central-banks-operational-framework-and-what-it-missing (Accessed: 14 August 2024).

Grandjean, A., Faure, J., Nyssen, J., Zaouati, P., Blanchet, E., Bour, D., Courderc, N., Defréville, H., Delpont, S., Demurger, P., France, P.-E., Guivarch, C., Huard, A., Hug de larauze, T., Leahu-Aluas, I., Lemaignan, B., Legentil, S., Mahfouz, R., Marchal, J., Monnet, É., Noé, J., Noldin, J., Philipponnat, T., Picard, F., Schubert, K., Touati, J., de Tourtier, A.-C. and Wolff, N. (2024) '« La BCE pourrait appliquer des taux d'intérêt moins élevés au refinancement des prêts bancaires alloués à la transition écologique »', *Le Monde.fr*, 2 March. Available at:

https://www.lemonde.fr/idees/article/2024/03/02/la-bce-pourrait-appliquer-destaux-d-interet-moins-eleves-au-refinancement-des-prets-bancaires-alloues-a-latransition-ecologique\_6219665\_3232.html (Accessed: 14 June 2024). Gwyn Jones, M. (2024) *Red Sea crisis a risk for EU economy, trade chief warns, euronews*. Available at: https://www.euronews.com/my-europe/2024/01/16/red-sea-crisis-a-risk-for-eu-economy-energy-supply-and-prices-trade-chief (Accessed: 5 September 2024).

de Haan, L., Holton, S. and van den End, J.W. (2019) *The impact of central bank liquidity support on banks' balance sheets*. ECB. Available at: https://www.ecb.europa.eu/pub/pdf/scpwps/ecb.wp2326~4f94de6a16.en.pdf.

IEA (2020) *Projected Costs of Generating Electricity, 2020 Edition*. Available at: https://iea.blob.core.windows.net/assets/ae17da3d-e8a5-4163-a3ec-2e6fb0b5677d/Projected-Costs-of-Generating-Electricity-2020.pdf.

IEA (2023) *World Energy Investment 2023*. Available at: https://iea.blob.core.windows.net/assets/8834d3af-af60-4df0-9643-72e2684f7221/WorldEnergyInvestment2023.pdf.

International Energy Agency (2023) *Renewable Energy Market Update - June 2023*. International Energy Agency. Available at: https://www.iea.org/reports/renewableenergy-market-update-june-2023.

Ioannidis, M., Hlásková, S.J. and Zilioli, C. (2021) 'The Mandate of the ECB: Legal Considerations in the ECB's Monetary Policy Strategy Review', *European Central Bank*, Occasional Papers. Available at: https://doi.org/10.2139/ssrn.3928298.

IRENA (2023) *Renewable Power Generation Costs in 2022*. International Renewable Energy Association. Available at: https://www.irena.org/Publications/2023/Aug/Renewable-Power-Generation-Costsin-2022 (Accessed: 27 May 2024).

Jakab, Z. and Kumhof, M. (2015) 'Banks are Not Intermediaries of Loanable Funds And Why this Matters', *Bank of England Staff Working Papers No. 529* [Preprint]. Available at: https://doi.org/10.2139/ssrn.2612050.

Kedward, K., Gabor, D. and Ryan-Collins, J. (2022) 'Aligning finance with the green transition: From a risk-based to an allocative green credit policy regime.', *SSRN Electronic Journal* [Preprint]. Available at: https://doi.org/10.2139/ssrn.4198146.

Kempf, H. (2020) 'Verdir la politique monétaire', *Revue d'économie politique*, Vol. 130(3), pp. 311–343. Available at: https://doi.org/10.3917/redp.303.0311.

Kleintop, J. (2023) 'What happened to ESG stocks?', 23 October. Available at: https://www.schwab.com/learn/story/what-happened-to-esgstocks#:~:text=Investments%20.

Krahé, M. and Heilmann, F. (2023) *Fossil fuel to the fire: Energy and inflation in Europe*. Research Report. Fachtexte. Available at: https://www.econstor.eu/handle/10419/279870 (Accessed: 28 June 2024).

Lagarde, C. (2022) 'Letter from Christine Lagarde to MEP Ville Niinistö'. Available at: https://www.ecb.europa.eu/pub/pdf/other/ecb.mepletter220706\_Niinisto~7a5c3268 3f.en.pdf (Accessed: 9 September 2024).

Lagarde, C. (2024) 'Policymaking in a new risk environment'. Dubrovnik, 14 June. Available at: https://www.ecb.europa.eu//press/key/date/2024/html/ecb.sp240614~baa283d69a.e n.html (Accessed: 28 June 2024).

Laine, O.-M. (2019) *The effect of TLTRO-II on bank lending*. Bank of Finland/Suomen Pankki. Available at: https://publications.bof.fi/bitstream/handle/10024/48041/BoF\_DP\_1907\_revision.pdf.

Laine, O.-M. and Nelimarkka, J. (2023) Assessing targeted longer-term refinancing operations: identification through search intensity. Bank of Finland/Suomen Pankki. Available at: https://publications.bof.fi/bitstream/handle/10024/53116/BoF\_DP\_2313.pdf.

Lane, T. (2024) Monetary Policy And Climate Change Distinguishing Between A Protective And A Proactive Agenda. Council on Economic Policies. Available at: https://www.cepweb.org/wp-content/uploads/2024/06/Lane-2024.-Monetary-Policy-and-Climate-Change.pdf.

Linzert, T., Nautz, D. and Bindseil, U. (2006) 'Bidding behavior in the longer term refinancing operations of the European Central Bank: Evidence from a panel sample selection model q'.

Lonergan, E. (2016) 'There is a lot more the ECB can do', *Philosophy of Money*, 23 February. Available at: https://www.philosophyofmoney.net/there-is-a-lot-more-the-ecb-can-do/ (Accessed: 18 June 2024).

Lonergan, E. (2019) 'Dual interest rates always work', *Philosophy of Money*, 5 June. Available at: https://www.philosophyofmoney.net/dual-interest-rates-always-work/ (Accessed: 18 June 2024).
73

Lonergan, E. (2020) 'European Central Bank has one item left in its toolkit: dual rates', 2 January. Available at: https://www.ft.com/content/885d0f9c-2319-11ea-92da-f0c92e957a96 (Accessed: 10 June 2024).

Lonergan, E., Ducrozet, F., Jourdan, S., Schoenmaker, D., Van Tilburg, R., Grünewald, S., Baarsma, B., Benink, H., Brzeski, C., Harreau, L., Phlippen, S. and Kounis, N. (2022) 'R.I.P. TLTROs, long live green dual rates'. Available at: https://sustainablefinancelab.nl/wp-content/uploads/sites/334/2022/06/Memo-ECB-June-5.pdf (Accessed: 11 June 2024).

Milligan, E. (2024) 'Trump Draws Up a Plan to Punish the EU Over Trade', *Bloomberg.com*, 7 February. Available at: https://www.bloomberg.com/news/newsletters/2024-02-07/donald-trump-plansto-punish-europe-over-trade (Accessed: 5 September 2024).

Millischer, L., Fy, C., Volz, U. and Beirne, J. (2024) 'Do Renewables Shield Inflation from Fossil Fuel-Price Fluctuations?', *IMF Working Papers*, 2024/111. Available at: https://www.imf.org/en/Publications/WP/Issues/2024/05/31/Do-Renewables-Shield-Inflation-from-Fossil-Fuel-Price-Fluctuations-549704#:~:text=In%20the%20wake%20of%20recently,fuel%20price%2Dinduced%20

inflation%20volatility.

Moens, B. and Gijs, C. (2023) *Trump's return strikes fear into the heart of Brussels*, *POLITICO*. Available at: https://www.politico.eu/article/donald-trump-europeanunion-reelection-trade-tariffs-steel-aluminum-fears/ (Accessed: 5 September 2024).

'Monetary Dialogue with the ECB President at the ECON Committee' (2020). Brussels. Available at: https://www.europarl.europa.eu/cmsdata/212180/CRE\_Monetary\_Dialogue\_2809202

0\_EN.pdf (Accessed: 4 June 2024).

Monnet, E., Humann, T. and Mitchener, K. (2024) 'Do Disinflation Policies Ravage Central Bank Finances?'

Monnet, É. and Klooster, J.V. (2023) Using green credit policy to bring down inflation- what central bankers can learn from history. INSPIRE. Available at: https://www.inspiregreenfinance.org/wp-content/uploads/2023/07/INSPIRE-Sustainable-Central-Banking-Toolbox\_Policy-Briefing-13.pdf.

Monnin, P. (2015) *The Impact of Interest Rates on Electricity Production Costs.* Council on Economic Policies. Available at: https://www.cepweb.org/the-impact-ofinterest-rates-on-electricity-production-costs/. Muller, N.Z. (2021) 'On the Green Interest Rate.' National Bureau of Economic Research (Working Paper Series). Available at: https://doi.org/10.3386/w28891.

NBB (2024) *Monetary plumbing*. Available at: https://www.nbb.be/en/blog/monetary-plumbing (Accessed: 9 September 2024).

NGFS (2021) Adapting central bank operations to a hotter world: Reviewing some options. NGFS. Available at:

https://www.ngfs.net/sites/default/files/media/2021/06/17/ngfs\_monetary\_policy\_op erations\_final.pdf.

NGFS (2024) Adapting central bank operations to a hotter world: current progress and insights from practical examples. Network For Greening the Financial System. Available at:

https://www.ngfs.net/sites/default/files/medias/documents/ngfs\_adapting\_central\_ bank\_operations\_to\_a\_hotter\_world\_final.pdf (Accessed: 11 July 2024).

NVDE (2023a) Assessment effects high interest rate for sustainable energy sector. Available at: https://www.nvde.nl/wp-content/uploads/2023/06/NVDE-Engelseversie-ledeninventarisatie-rapport-def-1.pdf.

NVDE (2023b) 'Lower interest rates for sustainable energy help curb inflation', *NVDE*, 14 June. Available at: https://www.nvde.nl/wpcontent/uploads/2023/06/persbericht-NVDE-hoge-rente-en-duurzameenergiesector-Engelse-def-1.pdf (Accessed: 17 June 2024).

Oustry, A., Erkan, B., Svartzman, R. and Weber, P.-F. (2020) *Climate-related Risks and Central Banks' Collateral Policy: a Methodological Experiment*. Banque de France.

Partington, Ri. (2024) 'Oil prices hit \$80 a barrel as fears grow over Red Sea disruption to trade', *The Guardian*, 12 January. Available at: https://www.theguardian.com/business/2024/jan/12/oil-prices-fears-grow-over-redsea-disruption-to-trade-yemen.

PBOC (2021) PBOC Officials Answer Press Questions on the Launch of Carbon Emission Reduction Facility, People's Bank of China. Available at: http://www.pbc.gov.cn/en/3688229/3688299/3688308/4385447/index.html (Accessed: 2 September 2024).

PBoC (2021) 'The People's Bank of China Launches the Carbon Emission Reduction Facility'. Available at: http://www.pbc.gov.cn/en/3688110/3688172/4157443/4385345/index.html. Pellerin-Carlin, T., Calipel, C. and Bizien, A. (2024) *European Climate Investment Deficit report: an investment pathway for Europe's future*. I4CE - Institute for Climate Economics. Available at: https://www.i4ce.org/en/publication/europeanclimate-investment-deficit-report-investment-pathway-europe-future/ (Accessed: 14 June 2024).

Polzin, F., Sanders, M., Steffen, B., Egli, F., Schmidt, T.S., Karkatsoulis, P., Fragkos, P. and Paroussos, L. (2021) 'The effect of differentiating costs of capital by country and technology on the European energy transition', *Climatic Change*, 167(1–2), p. 26. Available at: https://doi.org/10.1007/s10584-021-03163-4.

Quintana, J. (2024) 'The impact of renewable energies on wholesale electricity prices', *Bank of Spain Economic Bulletin*, (2024/Q3), p. 09. Available at: https://doi.org/10.53479/37635.

Reclaim Finance (2023) *Greening the Eurosystem Collateral Framework*. Available at: https://reclaimfinance.org/site/wp-content/uploads/2024/01/2023.11.06\_Reclaim-Finance\_Greening-Eurosystem-Collateral-Framework-1.pdf (Accessed: 6 September 2024).

Schmidt, T.S., Steffen, B., Egli, F., Pahle, M., Tietjen, O. and Edenhofer, O. (2019) 'Adverse effects of rising interest rates on sustainable energy transitions', *Nature Sustainability*, 2(9), pp. 879–885. Available at: https://doi.org/10.1038/s41893-019-0375-2.

Schnabel, I. (2022) 'A new age of energy inflation: climateflation, fossilflation and greenflation'. *The ECB and its Watchers XXII*, Frankfurt, 17 March. Available at: https://www.ecb.europa.eu/press/key/date/2022/html/ecb.sp220317\_2~dbb3582f0a. en.html (Accessed: 17 April 2024).

Schnabel, I. (2023) 'Monetary policy tightening and the green transition'. International Symposium on Central Bank Independence, Sveriges Riksbank, Stockholm, 10 January. Available at: https://www.ecb.europa.eu/press/key/date/2023/html/ecb.sp230110~21c89bef1b.en.h tml (Accessed: 4 June 2024).

Schnabel, I. (2024) 'The Eurosystem's operational framework'. *Money Market Contact Group meeting*, 14 March. Available at: https://www.ecb.europa.eu/press/key/date/2024/html/ecb.sp240314~8b609de772.e n.html (Accessed: 5 July 2024). Tamez, M., Weenink, H. and Yoshinaga, A. (2024) 'Central Banks and Climate Change: Key Legal Issues', *IMF Working Papers* [Preprint], (No. 2024/192). Available at: https://www.imf.org/en/Publications/WP/Issues/2024/09/10/Central-Banks-and-Climate-Change-Key-Legal-Issues-553517.

van Tilburg, R. and Simić, A. (2021) *Legally green: Climate change and the ECB mandate*. Sustainable Finance Lab. Available at: https://sustainablefinancelab.nl/wp-content/uploads/sites/334/2021/07/Legally-Green.pdf.

Van Doorslaer, H. and Vermeiren, M. (2023) 'Beyond normal central banking? Monetary policy after the pandemic', *SSRN Electronic Journal* [Preprint]. Available at: https://doi.org/10.2139/ssrn.4329228.

Van 't Klooster, J. and Weber, I. (2024) 'Closing the EU's inflation governance gap: The limits of monetary policy and the case for a new policy framework for shockflation', *European Parliament*, PE 755.727. Available at: https://doi.org/10.2861/712132.

Van 't Klooster, J. and De Boer, N. (2021) *The ECB's neglected secondary mandate: An inter-institutional solution*. Positive Money Europe. Available at: https://www.positivemoney.eu/wp-content/uploads/2021/10/The-ECBs-neglectedsecondary-mandate\_v6.0.pdf (Accessed: 30 April 2024).

Van 't Klooster, J. and Van Tilburg, R. (2020) *Targeting a sustainable recovery with Green TLTROs.* Positive Money Europe. Available at: https://www.positivemoney.eu/wp-content/uploads/2020/09/Green-TLTROs.pdf (Accessed: 10 April 2024).

Várhelyi, G. (2020) 'EU Taxonomy and the monetary policy prism', in. *ESCB Legal Conference 2020*, Frankfurt am Main, Germany: European Central Bank, pp. 144– 165.

Villeroy de Galhau, F. (2023) 'Hearing of Mr. Villeroy de Galhau at the Assemblée Nationale''. Available at: https://www.assemblee-nationale.fr/dyn/16/comptesrendus/cion\_fin/l16cion\_fin2223048\_compte-rendu#.

Voldsgaard, A., Egli, F. and Pollitt, H. (2022) 'Can we avoid green collateral damage from rising interest rates?', 20 June. Available at: https://medium.com/iippblog/can-we-avoid-green-collateral-damage-from-rising-interest-rates-1259ea94c9ea#:~:text=We%20suggest%20that%20green%20collateral,tandem%20 with%20rising%20interest%20rates. Zakeri, B., Staffell, I., Dodds, P.E., Grubb, M., Ekins, P., Jääskeläinen, J., Cross, S., Helin, K. and Castagneto Gissey, G. (2023) 'The role of natural gas in setting electricity prices in Europe', *Energy Reports*, 10, pp. 2778–2792. Available at: https://doi.org/10.1016/j.egyr.2023.09.069.

Sustainable Finance Lab

This is a Sustainable Finance Lab publication www.sustainablefinancelab.nl